

Staff Working Paper/Document de travail du personnel 2016-31

Housing Market Dynamics and Macroprudential Policy



by Gabriel Bruneau, Ian Christensen and Césaire A. Meh

Bank of Canada staff working papers provide a forum for staff to publish work-in-progress research independently from the Bank's Governing Council. This research may support or challenge prevailing policy orthodoxy. Therefore, the views expressed in this paper are solely those of the authors and may differ from official Bank of Canada views. No responsibility for them should be attributed to the Bank.

Bank of Canada Staff Working Paper 2016-31

July 2016

Housing Market Dynamics and Macroprudential Policy

by

Gabriel Bruneau,¹ Ian Christensen² and Césaire A. Meh³

¹Financial Stability Department ²Financial Markets Department ³International Economic Analysis Department Bank of Canada Ottawa, Ontario, Canada K1A 0G9 <u>gbruneau@bankofcanada.ca</u> <u>ichristensen@bankofcanada.ca</u> <u>cmeh@bankofcanada.ca</u>

Acknowledgements

The authors would like to thank Amélie Clément, Jason Allen, Brian Peterson, Stefano Gnocchi, Luisa Lambertini, Jörn Tenhofen and Randall Morck and participants at seminars at the Bank of Canada and Université de Montréal, at the Canadian Economics Association Annual Conference 2013, at the Swiss National Bank Research Conference 2015, at the Conference on Network Models, Stress Testing and Other Tools for Financial Stability Monitoring and Macroprudential Policy Design Implementation at the Banco de Mexico, at the American Economic Association Annual Meeting 2016 and at the Conference on Macroprudential Regulation at the IFSC Dublin for their useful comments, suggestions and discussions. The authors would also like to thank Andrew Usher, Hongyu Xiao, Omar Abdelrahman and William Leung for their excellent research assistance.

Abstract

We perform an analysis to determine how well the introduction of a countercyclical loanto-value (LTV) ratio can reduce household indebtedness and housing price fluctuations compared with a monetary policy rule augmented with house price inflation. To this end, we construct a New Keynesian model in which a fraction of households borrow against the value of their houses and we introduce news shocks on housing demand. We estimate the model with Canadian data using Bayesian methods. We find that the introduction of news shocks can generate a housing market boom-bust cycle, the bust following unrealized expectations on housing demand. Our study also suggests that a countercyclical LTV ratio is a useful policy to reduce the spillover from the housing market to consumption, and to lean against news-driven boom-bust cycles in housing price and credit generated by expectations of future macroeconomic developments.

JEL classification: E31, E42, H23

Bank classification: Business fluctuations and cycles; Financial stability; Housing; Monetary policy framework; Transmission of monetary policy

Résumé

Nous procédons à une analyse visant à déterminer dans quelle mesure la mise en place d'un rapport prêt-valeur contracyclique peut réduire l'endettement des ménages et les variations des prix des logements comparativement à l'instauration d'une règle de politique monétaire tenant compte de la hausse des prix des logements. À cette fin, nous élaborons un modèle de type nouveau keynésien dans lequel une partie des ménages empruntent sur la valeur de leur propriété et nous appliquons des chocs anticipés, induits par de nouvelles informations, à la demande de logements. Nous estimons le modèle à l'aide de données canadiennes en recourant à des méthodes bayésiennes. Nous constatons que l'application de tels chocs peut générer un cycle d'expansion et de contraction du marché du logement, la contraction découlant d'attentes non remplies au regard de la demande de logements. Notre étude donne aussi à penser qu'un rapport prêt-valeur contracyclique constitue une mesure utile pour réduire les répercussions négatives du marché du logement sur la consommation et contrer les cycles d'expansion et de contraction des prix des logements et du crédit à l'habitation engendrés par les nouvelles informations et l'anticipation de l'évolution macroéconomique.

Classification JEL : E31, E42, H23

Classification de la Banque : Cycles et fluctuations économiques; Stabilité financière; Logement; Cadre de conduite de la politique monétaire; Transmission de la politique monétaire

Non-Technical Summary

We perform an analysis to determine how well the introduction of a countercyclical loan-to-value (LTV) ratio responding to credit-to-GDP or house price can reduce household indebtedness and housing price fluctuations compared with a monetary policy rule augmented with house price inflation.

The analysis is conducted in the context of a New Keynesian model in which a fraction of households borrow against the value of their houses and we introduce news shocks on housing demand and multi-period fixed-rate mortgage loans. News shocks can generate optimistic or pessimistic expectations, and cause fluctuations in housing investment. Optimistic expectations lead to excessive housing investment, thereby causing a boom in the housing market not based on fundamentals. Once the news shocks are found to be unrealized, buyers revert their actions and a bust in the housing market follows. Therefore, our model allows us to study the effect of macroprudential policies in a context where agents can have (over-)optimistic expectations about the future and react by contracting more debt under fixed long-term contracts.

We estimate the model with Canadian data using Bayesian methods. We then assess the (in)effectiveness of leaning using an expanded Taylor rule compared with the countercyclical LTV policies.

We find that the introduction of news shocks can generate a housing market boombust cycle, the bust following unrealized expectations on housing demand. Our study also suggests that a countercyclical LTV ratio is a useful policy to reduce the spillover from the housing market to consumption and to lean against news-driven boom-bust cycles in housing price and credit generated by expectations of future macroeconomic developments.

1 Introduction

The correlation between consumption expenditures and house price over the business cycles is well documented in macroeconomic studies. Indeed, time-series estimates for a variety of countries — including Canada — have shown that the two variables tend to move together. Understanding the dynamics between house price and the accumulation of household debt is particularly important for policy-makers in designing and implementing public policy and regulation, as it has been established that housing busts preceded by large household debt increases tend to result in deeper recessions (IMF, 2012). As an example, the economic fallout resulting from the collapse of the U.S. housing market was more painful and prolonged relative to a standard recession, as households and financial institutions engaged in a long deleveraging process following the crisis. During the same expansionary period, Canada also experienced a significant increase in house price, residential mortgages and consumer credit.¹

In this paper, we perform an analysis to determine how well the introduction of a countercyclical loan-to-value (LTV) ratio² can reduce household indebtedness and housing price fluctuations compared with a monetary policy rule augmented with house price inflation. To this end, we construct a New Keynesian model in which a fraction of households borrow against the value of their houses and we introduce news shocks on housing demand. We estimate the model with Canadian data using Bayesian methods. We find that the introduction of news shocks can generate a housing market boom-bust cycle, the bust following unrealized expectations on housing demand. Our study also suggests that a countercyclical LTV ratio is a useful policy to reduce the spillover from the housing market to consumption and to lean against news-driven

¹House price doubled and ratios of house-price-to-income and house-price-to-rent increased sharply (IMF, 2013). Mortgage credit (including Home Equity Lines of Credit, or HELOCs) expanded by almost 9 percent per year on average between 2000 and 2008, while household debt as a share of disposable income rose from about 100 percent in 2000 to 165 percent in 2013. As a result, mortgage and consumer loans secured by real estate (mostly HELOCs) are estimated to account for 80 percent of household debt and to represent the single largest exposure for Canadian banks with about 35 percent of their assets.

 $^{^{2}}$ LTV ratio imposes a cap on the size of a mortgage loan relative to the value of a property, thereby requiring a minimum down payment.

boom-bust cycles in housing price and credit generated by expectations of future macroeconomic developments.

Our paper is related to the business cycle literature on the role of collateral constraints in the transmission of shocks. A non-exhaustive list includes the following studies. Iacoviello and Neri (2010) estimate a New Keynesian model and study the sources and consequences of fluctuations in the U.S. housing market. Their results suggest that slow technological progress in the housing sector explains the upward trend in real housing price over the last 40 years and that the housing market spillovers are non-negligible, concentrated on consumption rather than business investment, and have become more important over time. Monacelli (2009) incorporates a durable goods sector into a general equilibrium model with collateral constraint and examines the monetary policy reaction to sectoral fluctuations. Finally, Gelain, Lansing, and Mendicino (2013) find that the introduction of a simple moving-average forecast rule, a deviation from the rational expectations hypothesis, for a subset of agents can significantly magnify the volatility and persistence of house price and household debt relative to an otherwise similar model with fully rational expectations.

Our research is also related to papers that consider either or both the effects of monetary policy and changes in regulatory LTV in a dynamic stochastic general equilibrium (DSGE) framework similar to Iacoviello (2005) and Iacoviello and Neri (2010). A nonexhaustive list includes Christensen, Corrigan, Mendicino, and Nishiyama (2009), Kannan, Rabanal, and Scott (2012), Justiniano, Primiceri, and Tambalotti (2013), Lambertini, Mendicino, and Teresa Punzi (2013b), Gelain et al. (2013) and Gelain, Kolasa, and Brzoza-Brzezina (2014). Among them, Lambertini et al. (2013b) study the potential gains of monetary and macroprudential policies that lean against house price and credit cycles and find that, when the implementation of both interest-rate and LTV policies is allowed, heterogeneity in the welfare implications is key in determining the optimal use of policy instruments.

Our model shares many features with Iacoviello and Neri (2010). At the core of the

model is the borrowers-lenders set-up developed by Kiyotaki and Moore (1997). There are two types of households differentiated by the degree to which they discount the future. In equilibrium, one type of household is a lender and the other type a borrower. Borrowers face a collateral constraint that limits their ability to borrow to a fraction of the value of their housing assets. Rising house values can therefore improve the debt capacity of borrowers, allowing them to increase consumption. Households buy and sell housing in a centralized market. The mechanisms used in the paper are consistent with the view that credit, via collateral channel, is a propagation mechanism that causes macroeconomic fluctuations.

We extend the model of Iacoviello and Neri (2010) and contribute to the literature in two important dimensions. First, we introduce multi-period fixed-rate mortgage loans. Considering that the median length of a mortgage contract in Canada is 5 years and the majority are at a fixed rate, this feature is potentially crucial to replicate business cycle facts and to study the (in)effectiveness of macroprudential policies, assuming one-period loans as in Iacoviello and Neri (2010) and subsequent papers are not appropriate.³ Secondly, we introduce news shocks (Beaudry and Portier, 2004) on housing demand. The notion that optimistic expectations can cause housing market booms and busts is widely accepted by policy-makers. As shown in Shiller (2007), real housing price can significantly deviate from economic fundamentals, as during the 1998– 2007 housing boom. However, recent research on the housing sector in a DSGE model typically ignores expectation errors as a potential source of fluctuations. Notable exceptions are Kanik and Xiao (2014) and Lambertini et al. (2013b). News shocks can generate optimistic or pessimistic expectations and cause fluctuations in housing investment. Optimistic expectations lead to excessive housing investment, thereby causing a boom in the housing market not based on fundamentals. Once the news shocks are found to be unrealized, buyers revert their actions and a bust in the housing market follows. Therefore, our model allows us to study the effect of macroprudential

 $^{^{3}}$ This feature is introduced exogenously. Future research will study why this framework arises endogenously within the mortgage market.

policies in a context where agents can have (over-)optimistic expectations about the future and react by contracting more debt under fixed long-term contracts.

We estimate the model with Canadian data using Bayesian methods. We make three important contributions. First, we construct a new set of sectoral (disaggregated) data for consumption and the housing sector that allows us to better identify sectoral volatility and sector-specific dynamics. Second, we calibrated the share of patient and impatient households to reflect characteristics of wealth and income distribution data (Gelain et al., 2013). By using a calibration that underestimates the share of impatient households, recent research underestimates the mortgage-debt-to-GDP ratio, and thus results in the underestimation of the amplifier effects of macroprudential policy changes on the broader economy. Finally, in addition to the parameter priors, we also use the model priors, which better capture the dynamics of the model as a whole than the parameter priors alone.

Based on our model specifications, we find strong evidence suggesting important spillover effects between the housing market (and housing wealth creation) and the rest of the economy, and this link is mainly driven by demand and credit factors. We also find that the introduction of news shocks can generate a housing market boom-bust cycle, the bust following unrealized expectations on housing demand. In the news-driven business cycle literature, it is usually hard to generate the right co-movement among aggregate variables (Beaudry and Portier, 2004; Jaimovich and Rebelo, 2009). In our paper, two features of the model generate the co-movement: (i) the heterogeneity of households, bringing reallocation of housing stock between households, and (ii) the credit channel, by relaxing the collateral constraint of the borrowers, allowing them to borrow and consume more.

Finally, with the estimated model able to generate news-driven business cycles, we show that a countercyclical LTV ratio is a useful policy to reduce the spillover from the housing market to consumption and to lean against news-driven boom-bust cycles in housing price and credit generated by expectations of future macroeconomic developments.

The remainder of the paper is organized as follows. Section 2 presents the theoretical model. Section 3 describes the calibration, discusses estimation issues and our econometric strategy, and introduces the data employed. Section 4 discusses the estimation results and the overall performance of the model to describe business cycle characteristics, while Section 5 discusses the news-driven business cycles and reports the effect of the introduction of a countercyclical LTV ratio. Section 6 concludes.

2 A Model of Irreversible Housing Investment

We start from a standard New Keynesian set-up,⁴ extended to incorporate household heterogeneity, irreversible housing investment and credit frictions, as in Iacoviello (2005) and Iacoviello and Neri (2010). Our economic environment features heterogeneity among economic agents. We consider an economy populated by two types of households, designated as borrowers and lenders. Credit flows are generated by assuming ex ante heterogeneity in agents' subjective discount factors. Impatient agents (borrowers) differ from patient agents (lenders) in that they discount the future at a faster rate. Hence, in equilibrium, patient agents are net lenders while impatient agents are net borrowers. To prevent borrowing from growing at the desired limit instantaneously, we assume that borrowers face a credit constraint tied to the current value of their collateral. We depart from the usual set-up of one-period loans by allowing for multi-period loans with fixed interest rates (Gelain et al., 2014; Alpanda, Cateau, and Meh, 2014; Alpanda and Zubairy, 2014), which is a representation closer to the Canadian context where the most common mortgage loan contract has a length of 5 years and a fixed interest rate.

There are two sectors of production in the economy: consumption and housing. Each variety of consumption goods is produced by a single firm in a monopolistic competitive environment and its price is set in a staggered fashion \dot{a} la Calvo (1983). A

 $^{^{4}}$ We consider as *standard New Keynesian set-up* models incorporating households and firms having rational expectations, with a variety of market failures, among them imperfect competition.

representative firm produces houses in a perfectly competitive environment. Households supply differentiated labour in a monopolistic competitive environment, their wages being set in a staggered fashion $\dot{a} \, la$ Calvo (1983). They buy goods, deriving their utility from consumption goods and services provided by their housing stock. Credit flows are generated via perfectly competitive financial intermediaries, which accept deposits from patient households to lend to impatient households.⁵ Finally, a central bank conducts monetary policy according to a Taylor-type rule.

2.1 Households

Households $i \in \{P, I\}$, patient and impatient respectively, derive in period t utility from consumption goods $c_{i,t}$ and from services provided by their housing stock $h_{i,t}$. They supply labour and derive a disutility from hours worked in the consumption sector $n_{i,t}^c$, and in the housing sector $n_{i,t}^h$. They maximize their expected lifetime utility:

$$E_0 \sum_{t=0}^{\infty} \beta_i^t \epsilon_t^b U\left(c_{i,t}, h_{i,t}, n_{i,t}^c, n_{i,t}^h\right), \qquad (1)$$

where $\beta_i \in (0, 1)$ is the subjective discount factor and ϵ_t^b represents an exogenous process on discount rates that affects the intertemporal substitution of households (Smets and Wouters, 2007; Justiniano, Primiceri, and Tambalotti, 2010). The functional form of U is

$$U(\bullet) = \frac{(x_{i,t})^{1-\sigma_i^x}}{1-\sigma_i^x} - \frac{\epsilon_t^n}{1+\eta_i} \left(\left(n_{i,t}^c \right)^{\frac{\theta_i^n+1}{\theta_i^n}} + \left(n_{i,t}^h \right)^{\frac{\theta_i^n+1}{\theta_i^n}} \right)^{\frac{\theta_i^{r'}(1+\eta_i)}{\theta_i^n+1}},$$

where σ_i^x is the constant elasticity of substitution (CES), ϵ_t^n is an exogenous process on labour supplies, and θ_i^n and η_i are, respectively, the intratemporal elasticity of substitution between sectoral labour supplies and the inverse of Frish elasticity of substitution. This specification of disutility of labour follows Horvath (2000) and allows for imperfect labour mobility across sectors by making the agents less responsive

⁵The presence of financial intermediaries is not crucial to our model, as we could have allowed the patient households to lend directly to impatient households. However, first, it simplifies the exposition of the model as the aggregation of deposits and loans is done at the financial intermediary level, and not by the households, and second, it will be useful to have that feature for future extension of the model.

to sectoral wage differentials.⁶ The final good $x_{i,t}$ is defined as a CES composite of consumption goods $c_{i,t}$ and housing stock $h_{i,t}$:

$$x_{i,t} = \left[\left(1 - \epsilon_t^h \right)^{\frac{1}{\theta_i^x}} c_{i,t}^{\frac{\theta_i^x - 1}{\theta_i^x}} + \left(\epsilon_t^h \right)^{\frac{1}{\theta_i^x}} h_{i,t}^{\frac{\theta_i^x - 1}{\theta_i^x}} \right]^{\frac{\theta_i^z}{\theta_i^x - 1}},$$
(2)

where ϵ_t^h is an exogenous process on the preference for services provided by the housing stock (housing demand shock). Housing demand shock includes both unanticipated (surprise) and anticipated (news) components, while all the other shocks in the model include unanticipated shocks only. We focus exclusively on studying news shocks on demand exogenous processes, as the effects of news shocks on supply exogenous processes have been covered extensively in Lambertini et al. (2013b). More details are provided in Section 2.5. The parameter θ_i^x is the intratemporal elasticity of substitution between consumption goods and the services provided by the housing stock.⁷ This formulation of utility is slightly different than the standard utility in this literature, as we differ from the log-log specification of Iacoviello (2005) and Iacoviello and Neri (2010). Our specification tests. Households accumulate housing stock according to the law of motion

$$h_{i,t} = (1 - \delta^h) h_{i,t-1} + i^h_{i,t}, \qquad (3)$$

where $i_{i,t}^{h}$ is the investment in housing stock and δ^{h} its fixed depreciation rate. Aggregate housing investment is irreversible, while the housing investment at the household level is reversible at the real equilibrium house price. The irreversibility at the aggregate level is introduced via a specific production sector for housing goods, in which the production cannot be negative.⁸ Therefore the housing stock can decrease at a

 $^{{}^{6}\}theta_{i}^{n} \in [0, \infty)$, and when $\theta_{i}^{n} \to \infty$, hours worked in each sector tends to be a perfect substitute as agents devote all of their time to the sector paying the highest wage and all sectors pay the same hourly wage at the margin. Another way to include imperfect mobility between sectors would be the introduction of labour adjustment cost at the firm level.

 $^{{}^{7}\}theta_{i}^{x} \in [0,\infty)$, and when $\theta_{i}^{x} \to \infty$, both goods tend to be perfect substitutes, whereas they tend to be perfect complements when $\theta_{i}^{x} \to 0$.

⁸Hours worked in the housing sector cannot be negative by the equilibrium conditions.

maximum rate of $\delta^{h.9}$

Labour Labour decisions are made by a central authority within the households, which supplies, in a monopolistic competitive environment, differentiated labour $n_{i,e,t}^{j}$ in a continuum of labour markets $e \in [0, 1]$ in sector $j \in \{c, h\}$ (Erceg, Henderson, and Levin, 2000; Schmitt-Grohe and Uribe, 2007).¹⁰ Both sectors are, in terms of notation, the same. The central authority supplies labour to satisfy the demand given by

$$n_{i,e,t}^{j} = \left(\frac{W_{i,e,t}^{j}}{W_{i,t}^{j}}\right)^{-\theta^{n^{j}}} n_{i,t}^{j,d},$$
(4)

where $W_{i,e,t}^{j}$ denotes the nominal wage charged by the central authority in the labour market e in sector j for agents of type i, $W_{i,t}^{j}$ is the nominal wage index, $n_{i,t}^{j,d}$ is a measure of aggregate labour demand by firms and $\theta^{n^{j}}$ is the wage-elasticity of demand.¹¹ In each labour market, the central authority takes $W_{i,t}^{j}$ and $n_{i,t}^{j,d}$ as given. In addition, the total number of hours allocated to the different labour markets must satisfy the resource constraint in each sector

$$n_{i,t}^{j} = \int_{0}^{1} \left(n_{i,e,t}^{j} \right) de.$$
(5)

Combining this restriction with equation (4) yields the aggregated labour supply, expressed in real terms, in each sector j:

$$n_{i,t}^{j} = n_{i,t}^{j,d} \int_{0}^{1} \left(\frac{w_{i,e,t}^{j}}{w_{i,t}^{j}}\right)^{-\theta^{n^{j}}} de.$$
(6)

⁹The Canadian data show that the value of house demolished (i.e. disinvestment in housing stock) over the total stock of house is extremely small.

 $^{^{10}}$ By assuming that all households will act as a representative household, this set-up avoids the need to assume separability of preferences and the existence of insurance for labour markets. This feature is already reflected in the notation since there are no subscripts for the continuum of households of each type i.

¹¹The formal derivation of the labour demand by firms is presented in Section 2.2.

Patient Patient households (i = P) have a higher propensity to save (i.e. $\beta_P > \beta_I$). In equilibrium, they supply loans to impatient households (i = I) via their deposit d_t at financial intermediaries and accumulate housing and capital stock. Since lenders are the owners of the banks and firms in both sectors, they receive dividends f_t^c , f_t^h and f_t^{fi} from the consumption and housing sectors and from financial intermediaries, respectively. They maximize their expected lifetime utility (1) subject to their budget constraint in real terms

$$c_{P,t} + q_t^h i_{P,t}^h + \sum_{j \in \{c,h\}} q_t^{k^j} i_t^{k^j} + q_t^l l_t + b_{P,t} + d_t = \sum_{j \in \{c,h\}} r_t^{k^j} u_t^{k^j} k_{t-1}^j + \left(q_t^l + r_t^l\right) l_{t-1} +$$
(7)
$$\sum_{j \in \{c,h\}} n_{P,t}^{j,d} \int_0^1 w_{P,e,t}^j \left(\frac{w_{P,e,t}^j}{w_{P,t}^j}\right)^{-\theta^{n^j}} de + \frac{R_{t-1}b_{P,t-1}}{\pi_t^c} + \sum_{j \in \{c,h,fi\}} f_t^j + \frac{1}{\phi^m} \sum_{s=1}^{\phi^m} \frac{d_{t-s}}{\Pi_{v=-s}^0 \pi_{t+v}^c} + \sum_{s=1}^{\phi^m} \left(R_{t-s}^d - 1\right) \left(\frac{\phi^m - s + 1}{\phi^m}\right) \frac{d_{t-s}}{\Pi_{v=-s}^0 \pi_{t+v}^c},$$

the law of motion for capital in sector j

$$k_t^j = \left(1 - \delta_t^{k^j}\right) k_{t-1}^j + z_t^{i^k} i_t^{k^j} \left[1 - \frac{\phi^{k^j}}{2} \left(\frac{i_t^{k^j}}{i_{t-1}^{k^j}} - 1\right)^2\right],\tag{8}$$

and the law of motion for housing stock (3), where q_t^h is the real price of housing and $i_{P,t}^h$ is the investment in housing stock. Further, $j \in \{c, h\}$, k_t^j is the stock of capital specific to sector j, $i_t^{k^j}$ its investment level, $r_t^{k^j}$ its real rental rate, $u_t^{k^j}$ its variable capacity utilization rate and $\delta_t^{k^j}$ its variable depreciation rate. Lenders face an adjustment cost while investing in capital, parameterized by ϕ^{k^c} and ϕ^{k^h} . Moreover, the technology transforming investment goods into capital goods is subject to a transitory exogenous process denoted $z_t^{i^k}$ (Justiniano et al., 2010), and then induces a time-varying real price of investment $q_t^{k^j}$. Lenders own all the land stock l_t , which has a real price q_t^l and a real rental rate r_t^l . The stock of land is exogenous and fixed. π_t^c is the gross inflation rate in the consumption sector and R_t is the gross nominal interest rate on riskless one-period bonds $b_{P,t}$. Lenders' savings take the form of long-term deposits d_t at the financial intermediaries at the fixed interest rate R_t^d . As these deposits are the only source of funding for the financial intermediaries to finance the long-term loans, we have to impose long-term deposits on patient households to ensure zero profit conditions for the financial intermediaries. Finally, the deposit length is ϕ^m periods and, at each period, the lenders receive a share $\frac{1}{\phi^m}$ of the principal as a reimbursement of the deposit and a fixed return on investment $(R_t^d - 1)$ on the principal not reimbursed at the last period.

Lenders can control the intensity at which the capital stock is utilized. The effective amount of capital services supplied to firms in the consumption and housing sectors is given by $u_t^{k^c}k_{t-1}^c$ and $u_t^{k^h}k_{t-1}^h$, respectively. We assume that increasing the intensity of capital utilization entails a cost in the form of a faster rate of depreciation. Specifically, we assume that depreciation rate $\delta_t^{k^j}$ is an increasing and convex function of the rate of capacity utilization

$$\delta_t^{k^j} = \delta_0^{k^j} + \delta_1^{k^j} \left(u_t^{k^j} - 1 \right) + \frac{\delta_2^{k^j}}{2} \left(u_t^{k^j} - 1 \right)^2, \tag{9}$$

with $\delta_0^{k^j}, \delta_1^{k^j}, \delta_2^{k^j} > 0$, as in Schmitt-Grohe and Uribe (2012).¹²

Impatient The impatient households (i = I) do not accumulate physical capital nor hold any equity, and have access to multi-period fixed-rate mortgage loans with fixed (linear) principal payments, so that in each period borrowers have to pay interest on the outstanding debt and repay the amount of principal due. They maximize their expected lifetime utility (1) subject to a budget constraint

$$c_{I,t} + q_t^h i_{I,t}^h + b_{I,t} + \frac{1}{\phi^m} \sum_{s=1}^{\phi^m} \frac{m_{t-s}}{\Pi_{v=-s}^0 \pi_{t+v}^c} + \sum_{s=1}^{\phi^m} \left(R_{t-s}^m - 1 \right) \left(\frac{\phi^m - s + 1}{\phi^m} \right) \frac{m_{t-s}}{\Pi_{v=-s}^0 \pi_{t+v}^c} = \sum_{j \in \{c,h\}} n_{I,t}^{j,d} \int_0^1 w_{I,e,t}^j \left(\frac{w_{I,e,t}^j}{w_{I,t}^j} \right)^{-\theta^{n^j}} de + \frac{R_{t-1}b_{I,t-1}}{\pi_t^c} + m_t, \quad (10)$$

 $^{^{12}}$ See Appendix A for the Lagrangian and the complete set of first-order necessary conditions.

to the law of motion for housing stock (3), and a borrowing constraint. Private borrowing is subject to an endogenous limit. At any time t, borrowers agree to borrow no more than a share ω (Kiyotaki and Moore, 1997; Iacoviello, 2005; Monacelli, 2009; Iacoviello and Neri, 2010) of the current value of their housing stock:

$$M_t \ge -\omega \epsilon_t^{\chi} q_t^h h_{i,t},\tag{11}$$

where ϵ_t^{χ} is an exogenous process on credit, and

$$M_{t} = \sum_{s=0}^{\phi^{m}-1} \left(\frac{\phi^{m}-s}{\phi^{m}}\right) \frac{m_{t-s}}{\prod_{v=-s}^{0} \pi_{t+v}^{c}},$$
(12)

is the total mortgage debt.¹³ The model reflects the fact that mortgage debt is reoptimized only for the share of contracts that reach their end and must be refinanced. This type of long-term loan has only just begun to be studied in the literature (Gelain et al., 2014; Alpanda et al., 2014; Alpanda and Zubairy, 2014).¹⁴

Wages We introduce wage stickiness by assuming that, in each period, the central authority within the households *i* cannot set the nominal wage optimally for a share $\xi^{w^j} \in (0, 1)$ of labour markets chosen randomly. In the ξ^{w^j} labour markets that cannot set wages optimally, the wages are imperfectly indexed at rate ι^{w^j} to the steady-state inflation and at rate $1 - \iota^{w^j}$ to the t - 1 inflation. The reoptimization probability is independent and identically distributed (i.i.d.) across labour markets and over time. In labour markets in which the wage rate is reoptimized in period t, the real wage is set to equate the expected future average marginal revenue to the average marginal cost of supplying labour, with $\frac{\theta^{n^j}}{\theta^{n^j-1}}$ being the markup of wages over the marginal cost of labour that would prevail in the absence of wage stickiness and trend inflation. $\tilde{w}_{i,t}^{i}$ denotes the real wage prevailing in the $\left(1 - \xi^{w^j}\right)$ labour markets in which the central authority can set wages optimally in sector j in period t. Because the labour demand curve faced by the union is identical across all labour markets, and because

 $^{^{13}\}mathrm{We}$ use the convention that m<0 is a debt.

¹⁴See Appendix A for the Lagrangian and the complete set of first-order necessary conditions.

the cost of supplying labour is the same for all markets, one can assume that wage rate $\tilde{w}_{i,t}^{j}$ is identical for all industries within a given sector (but not necessarily across sectors).¹⁵

2.2 Firms

2.2.1 Consumption Sector

Final-Goods Producers Perfectly competitive final-consumption-goods producers purchase differentiated intermediate goods $m \in [0, 1]$ to assemble final goods y_t^c via the Dixit-Stiglitz aggregator

$$y_t^c = \left[\int_0^1 \left(y_{m,t}^c\right)^{\frac{\theta^c - 1}{\theta^c}} dm\right]^{\frac{\theta^c}{\theta^c - 1}},\tag{13}$$

where parameter θ^c denotes the intratemporal elasticity of substitution across varieties of intermediate differentiated goods¹⁶ and $y_{m,t}^c$ is the demand for goods of variety m. When maximizing their profits, final-goods producers take as given the prices of intermediate goods and the aggregate price index.¹⁷ The demand for goods of variety m is then given by

$$y_{m,t}^c = \left(\frac{P_{m,t}^c}{P_t^c}\right)^{-\theta^c} y_t^c,\tag{14}$$

where $P_{m,t}^c$ denotes the price of the intermediate consumption good m and P_t^c is the nominal price index defined as

$$P_t^c = \left[\int_0^1 \left(P_{m,t}^c\right)^{1-\theta^c} dm\right]^{\frac{1}{1-\theta^c}}.$$
(15)

¹⁵Part of the optimization problem (i.e. the Lagrangian) that is relevant for this purpose along with the first-order necessary condition can be found in Appendix A.

¹⁶When $\theta^c \to 0$, intermediate goods are perfect complements, whereas they are perfect substitutes when $\theta^c \to \infty$.

 $^{^{17}\}mathrm{The}$ optimization problem can be found in Appendix A.

Intermediate-Goods Producers Each variety of intermediate goods in the consumption sector is produced by a single firm m evolving in a monopolistic competitive environment. The production function for each of these firms $m \in [0, 1]$ is

$$y_{m,t}^{c} = z_{t}^{c} \left(k_{m,t}^{c}\right)^{\gamma^{c}} \left(\left(n_{P,m,t}^{c,d}\right)^{\alpha^{c}} \left(n_{I,m,t}^{c,d}\right)^{1-\alpha^{c}}\right)^{1-\gamma^{c}},$$
(16)

where $y_{m,t}^c$ is its total production, $n_{P,m,t}^{c,d}$ and $n_{I,m,t}^{c,d}$ are the number of hours of work demanded by the firm for both types of workers, z_t^c is the sector-wide total factor productivity and $k_{m,t}^c$ is the capital stock rented by the firm. Also, γ^c is the capital share of income and α^c is the lenders' share of labour income. We assume that the firm must satisfy the aggregated demand for good m (14) at posted price.¹⁸

Firms are able to reoptimize their prices as in Calvo (1983) and Yun (1996). Specifically, each firm faces a price rigidity with a non-zero probability ξ^{p^c} of being unable to adjust its nominal price in a given period. These firms are able to imperfectly index their price at rate ι^{p^c} to the steady-state inflation and at rate $1 - \iota^{p^c}$ to the t - 1inflation. The reoptimization probability is i.i.d. across firms and over time. Firms maximize the expected present value of their real dividends. Therefore, in setting their price in period t, firms take into account the fact that they may have to wait some time until they are able to reoptimize their price. In particular, the probability of not reoptimizing between dates t and t + s is $(\xi^{p^c})^s$. Since all reoptimizing firms face the same problem, they will all choose \tilde{p}_t^c as the real optimal price. Optimizing firms set nominal prices so that average future expected marginal revenues equate average future expected marginal costs.¹⁹

Given that the opportunity to reoptimize prices arrives probabilistically for each firm in each period, the aggregate price index (15) can be written in real terms in this

¹⁸The optimization problem along with the first-order necessary condition for wages and rental rate can be found in Appendix A.

¹⁹Part of the optimization problem (i.e. the Lagrangian) that is relevant for this purpose along with the first-order necessary condition can be found in Appendix A.

recursive form:

$$1 = \left(1 - \xi^{p^{c}}\right) \left(\tilde{p}_{t}^{c}\right)^{1-\theta^{c}} + \xi^{p^{c}} \left(\frac{\left(\pi^{c}\right)^{\iota^{p^{c}}} \left(\pi_{t-1}^{c}\right)^{1-\iota^{p^{c}}}}{\pi_{t}^{c}}\right)^{1-\theta^{c}}.$$
(17)

2.2.2 Housing Sector

A representative firm produces houses in a perfectly competitive environment. Its production function is

$$y_{t}^{h} = z_{t}^{h} \left(u_{t}^{k^{h}} k_{t-1}^{h} \right)^{\gamma^{h}} l_{t-1}^{\gamma^{l}} \left(\left(n_{P,t}^{h,d} \right)^{\alpha^{h}} \left(n_{I,t}^{h,d} \right)^{1-\alpha^{h}} \right)^{1-\gamma^{h}-\gamma^{l}},$$
(18)

where y_t^h is the total production for the housing sector, $n_{i,t}^{h,d}$ are the number of hours of work demanded by the firm for both types of workers, z_t^h is the sector-wide total factor productivity, $u_t^{k^h}k_{t-1}^h$ is the capital stock rented, and l_{t-1} is the land stock rented.²⁰ Also, γ^h is the capital share of income, α^h is the lenders' share of labour income and γ^l is the land share of income.²¹ The fixed stock of land creates decreasing return to scale in the housing sector, as the availability of land has been identified in the literature to be one of the drivers of the housing price increase over the last two decades in the major Canadian city areas, mainly Vancouver and Toronto.

2.2.3 Labour Input

The labour input used by the firms in a given sector, denoted by $n_{i,t}^{j,d}$, is assumed to be a composite made of a continuum of differentiated labour services $n_{i,e,t}^{j,d}$. In the case of the consumption sector, we first need to integrate labour demand over all

²⁰Since we assume a representative firm, to simplify the notation, we have already included the fact that the firm rents all the available capital in the sector, represented by $u_t^{k^h} k_{t-1}^h$, and all the available land, represented by l_{t-1} .

²¹In the calibration, we will assume that $\alpha^h = \alpha^c = \alpha$. The optimization problem along with the first-order necessary condition for wages and rental rate can be found in Appendix A.

intermediate firms $m \in [0, 1]$, which yields

$$n_{i,e,t}^{c,d} = \int_{0}^{1} \left(n_{i,e,m,t}^{c,d} \right) dm.$$
(19)

The aggregated labour demand for agents of type i in sector j is given by

$$n_{i,t}^{j,d} = \left[\int_{0}^{1} \left(n_{i,e,t}^{j,d} \right)^{\frac{\theta^{n^{j}}-1}{\theta^{n^{j}}}} de \right]^{\frac{\theta^{n^{j}}}{\theta^{n^{j}}-1}},$$
(20)

where $n_{i,e,t}^{j,d}$ is the demand for labour input of type e. When minimizing the cost, firms take as given the wage of labour input and the aggregate wage index. The optimal demand is

$$n_{i,e,t}^{j} = \left(\frac{W_{i,e,t}^{j}}{W_{i,t}^{j}}\right)^{-\theta^{n^{j}}} n_{i,t}^{j,d},$$
(21)

where the nominal wage index is given by

$$W_{i,t}^{j} = \left[\int_{0}^{1} \left(W_{i,e,t}^{j} \right)^{1-\theta^{n^{j}}} de \right]^{\frac{1}{1-\theta^{n^{j}}}}.$$
 (22)

Given that the opportunity to reoptimize wages arrives probabilistically for each household in each period, the aggregate wage index (22) can be written in real terms in this recursive form:

$$(w_{i,t}^{j})^{1-\theta^{n^{j}}} = \left(1-\xi^{w^{j}}\right) \left(\tilde{w}_{i,t}^{j}\right)^{1-\theta^{n^{j}}} + \xi^{w^{j}} \left(\frac{w_{i,t-1}^{j} \left(\pi^{c}\right)^{\iota^{w^{j}}} \left(\pi^{c}_{t-1}\right)^{1-\iota^{w^{j}}}}{\pi^{c}_{t}}\right)^{1-\theta^{n^{j}}}.$$
(23)

2.3 Financial Intermediaries

We assume that households use financial intermediaries because they cannot borrow and lend with each other directly. As mentioned before, the presence of financial intermediaries is not crucial to our model, as we could have allowed the patient households to lend directly to impatient households. However, it simplifies the exposition of the model as the aggregation of deposits and loans is done at the financial intermediary level, and not by the households.²² Financial intermediaries accept deposit d_t from lenders at the cost R_t^d and lend to borrowers m_t at rate R_t^m . The spread between rates on loans and deposits reflects a time-varying intermediation cost and it is assumed to be a deadweight loss to the economy. As Canada is a small open economy, trying to model the Canadian interest rate spread in a closed-economy model would have been difficult, as a major source of spread is the bank's funding cost on foreign markets. Therefore, modelling it as an exogenous process simplifies the process while keeping the transmission channel to the real economy. Financial intermediaries are assumed to be perfectly competitive and maximize the expected present value of their real dividends²³ subject to their balance sheets (in real terms)

$$d_t + \frac{1}{\phi^m} \sum_{s=1}^{\phi^m} \frac{m_{t-s}}{\prod_{v=-s}^0 \pi_{t+v}^c} + \sum_{s=1}^{\phi^m} \left(R_{t-s}^m - 1 \right) \left(\frac{\phi^m - s + 1}{\phi^m} \right) \frac{m_{t-s}}{\prod_{v=-s}^0 \pi_{t+v}^c} =$$
(24)

$$m_t + \frac{1}{\phi^m} \sum_{s=1}^{\phi^m} \frac{d_{t-s}}{\prod_{v=-s}^0 \pi_{t+v}^c} + \sum_{s=1}^{\phi^m} \left(R_{t-s}^d - 1 \right) \left(\frac{\phi^m - s + 1}{\phi^m} \right) \frac{d_{t-s}}{\prod_{v=-s}^0 \pi_{t+v}^c} + f_t^{fi} + \epsilon_t^{R^m} m_t.$$

As documented by Campbell (2013), in many countries (including Canada), the vast majority of housing loans are long-term fixed-rate mortgages. We then incorporate this type of contract with one type of mortgage being available, with principal being reimbursed linearly over ϕ^m periods. It is obviously a simple modelling, abstracting from many mortgage characteristics, among them amortization, prepayment, availability of mortgage with variable rate, home equity line of credit, etc.²⁴ The deposit is also of the same form.²⁵

 $^{^{22}}$ It will also be useful to have that feature for future extension of the model.

 $^{^{23}}$ See the objective function in the Appendix.

²⁴Also, we impose this type of loan exogenously to all households, while it is chosen endogenously by 70 percent of the households, among different type of contracts.

²⁵This hypothesis is not central to our results, but ensures zero profit over the terms of mortgage as monetary and spread shocks do not have any impact on the cost of funding. We tried one-period variable-rate deposit and it yielded similar results, but created profits or loss over the cycle and then

Taking the first-order necessary conditions for d_t and m_t yields the solution for R_t^m

$$R_{t}^{m} = \frac{\left(1 + \epsilon_{t}^{R^{m}}\right) - \frac{1}{\phi^{m}} \sum_{s=0}^{\phi^{m}} \beta_{P}^{s} E_{t} \left[\frac{\epsilon_{t+s}^{b}}{\epsilon_{t}^{b}} \frac{\lambda_{P,t+s}^{c}}{\lambda_{P,t}^{c}} \frac{1}{\prod_{v=1}^{s} \pi_{t+v}^{c}}\right]}{\sum_{s=0}^{\phi^{m}} \frac{\phi^{m} - s + 1}{\phi^{m}} \beta_{P}^{s} E_{t} \left[\frac{\epsilon_{t+s}^{b}}{\epsilon_{t}^{b}} \frac{\lambda_{P,t+s}^{c}}{\lambda_{P,t}^{c}} \frac{1}{\prod_{v=1}^{s} \pi_{t+v}^{c}}\right]}$$

2.4 Monetary Policy

The central bank implements a Taylor-type (Taylor, 1993) monetary policy rule with interest smoothing:

$$R_{t} = \rho_{r}R_{t-1} + (1 - \rho_{r})\left(R + \rho_{\pi^{c}}\left(\frac{\prod_{v=1}^{4}\pi_{t+v}^{c}}{4} - \epsilon_{t}^{\pi^{c}}\right) + \rho_{y}\left(Y_{t} - Y\right)\right).$$
 (25)

The monetary authority adjusts the nominal gross interest rate R_t from its steadystate value in response to deviations of inflation π_t^c from its target, deviations of the GDP (Y_t) from its steady-state value, and the i.i.d. monetary policy innovation ε_t^R with variance $\sigma_{\varepsilon^R}^2$. ρ_r , ρ_{π^c} and ρ_y are the persistence parameter, and the inflation and output response parameters, respectively. The central bank's target, $\epsilon_t^{\pi^c}$, is assumed to be an exogenous time-varying process subject to shocks, as in Smets and Wouters (2003) and Adolfson, Laseen, Linde, and Villani (2007). The inflation targeting has been implemented since 1991 in Canada, therefore this model specification can help capture the response of R_t to movements in π_t^c in the first third of our sample.

2.5 Exogenous Processes

All the exogenous processes in the model introduced earlier follow

$$\ln \Theta_t = (1 - \rho_\Theta) \Theta + \rho_\Theta \Theta_{t-1} + \varepsilon_t^\Theta, \qquad (26)$$

would potentially create entry-exit dynamics.

where $\Theta_t = \left\{ \epsilon_t^{\chi}, \epsilon_t^b, \epsilon_t^n, \epsilon_t^{\pi^c}, \epsilon_t^{R^m}, z_t^c, z_t^h, z_t^{i^k} \right\}$ are the exogenous processes, Θ are their respective steady-state values, and ρ_{Θ} their respective persistence parameters. The structural shocks in the model, $\varepsilon_t^{\Theta} = \left\{ \varepsilon_t^{\epsilon_{\chi}}, \varepsilon_t^{\epsilon_{h}}, \varepsilon_t^{\epsilon_{h}}, \varepsilon_t^{\epsilon_{h}}, \varepsilon_t^{\epsilon_{h}^{\pi^c}}, \varepsilon_t^{\epsilon_{h}^{R^m}}, \varepsilon_t^{z_{h}^{c}}, \varepsilon_t^{z_{h}^{h}}, \varepsilon_t^{z_{h}^{i^k}} \right\}$, along with the monetary policy innovation ε_t^R , are all zero-mean i.i.d. shocks with processspecific variance σ_{Θ}^2 and are uncorrelated contemporaneously and at all leads and lags, except for the housing demand shock, where it is the individual components (unanticipated and anticipated shocks) that independently have these characteristics.

Housing Demand Shocks We assume that economic agents have in period t an information set that goes beyond current and past realizations of ε^{ϵ^h} . The housing demand innovation $\varepsilon^{\epsilon^h}_{t-i}$, $\forall i$ is now composed of unanticipated and anticipated components:

$$\varepsilon_{t-i}^{\epsilon^h} = \sum_{j=0,4,8}^s \varepsilon_{t-i-j}^{\epsilon^h,j}.$$

In this formulation, agents observe the current and past values of the housing demand news shocks $\varepsilon^{\epsilon^{h,j}}$. The notation $\varepsilon_{t-i}^{\epsilon^{h,j}}$, $\forall i, j$ means that the anticipated disturbances (or news shocks) learned in t - i will affect the economy in j periods ahead (i.e. we learn in t - i a news shock that will happen in t - i + j). More specifically, the disturbance $\varepsilon_t^{\epsilon^{h,4}}$ represents an innovation to ϵ_{t+4}^{h} , which is announced in period t but materializes only in period t + 4. Note that $\varepsilon_t^{\epsilon^{h,4}}$ does not appear in the expression for $\varepsilon_t^{\epsilon^{h}}$ given above. Rather, the above expression features $\varepsilon_{t-4}^{\epsilon^{h,4}}$, the four-periodsahead announcement made in period t - 4. Similarly, $\varepsilon_t^{\epsilon^{h,8}}$ would be observed in t and represent eight-periods-ahead announcements of future changes in the housing demand. In this set-up, $\varepsilon_t^{\epsilon^{h,0}}$ can be viewed as the usual contemporaneous (i.e. unanticipated) disturbances to ϵ_t^{h} .

Since agents are forward-looking, they use the information contained in the realizations of news shocks in their current choices of consumption, investment in capital and housing and hours worked. It is precisely this forward-looking behaviour of economics agents that allows us, via our econometric method, to identify the volatilities of housing demand news shocks, even if we cannot observe them directly.

2.6 Market Clearing

Consumption Sector The aggregation in the production sectors and labour markets follows similar processes introduced in the New Keynesian literature. Integrating both sides of the intermediate goods production technology (16) yields

$$\int_{0}^{1} (y_{m,t}^{c}) dm = \int_{0}^{1} z_{t}^{c} (k_{m,t}^{c})^{\gamma^{c}} \left(\left(n_{P,m,t}^{c,d} \right)^{\alpha^{c}} \left(n_{I,m,t}^{c,d} \right)^{1-\alpha^{c}} \right)^{1-\gamma^{c}} dm \qquad (27)$$
$$= z_{t}^{c} \left(u_{t}^{k^{c}} k_{t-1}^{c} \right)^{\gamma^{c}} \left(\left(n_{P,t}^{c,d} \right)^{\alpha^{c}} \left(n_{I,t}^{c,d} \right)^{1-\alpha^{c}} \right)^{1-\gamma^{c}}.$$

Substituting $y_{m,t}^c$ in (27) and using demand function (14), we get

$$\left[\int_{0}^{1} \left(\frac{P_{m,t}^{c}}{P_{t}^{c}}\right)^{-\theta^{c}} dm\right] y_{t}^{c} = s_{t}^{y} y_{t}^{c} = z_{t}^{c} \left(u_{t}^{k^{c}} k_{t-1}^{c}\right)^{\gamma^{c}} \left(\left(n_{P,t}^{c,d}\right)^{\alpha^{c}} \left(n_{I,t}^{c,d}\right)^{1-\alpha^{c}}\right)^{1-\gamma^{c}}, \quad (28)$$

where s_t^y captures the inefficiencies associated with price dispersion arising from the price rigidity. Schmitt-Grohe and Uribe (2007) show that these price dispersion indexes can be defined as

$$s_t^y = (1 - \xi^c) \left(\tilde{p}_t^c \right)^{-\theta^c} + \xi^c \left(\frac{\left(\pi^c \right)^{\nu^{p^c}} \left(\pi_{t-1}^c \right)^{1 - \nu^{p^c}}}{\pi_t^c} \right)^{-\theta^c} s_{t-1}^y.$$
(29)

The market clearing condition for the consumption sector is therefore

$$y_t^c = c_t + q_t^{k^c} i_t^{k^c} \left(1 + \frac{\phi^{k^c}}{2} \left(\frac{i_t^{k^c}}{i_{t-1}^{k^c}} - 1 \right)^2 \right) + q_t^{k^h} i_t^{k^h} \left(1 + \frac{\phi^{k^h}}{2} \left(\frac{i_t^{k^h}}{i_{t-1}^{k^h}} - 1 \right)^2 \right) + \epsilon_t^{R^m} m_t,$$
(30)

where $c_t = c_{P,t} + c_{I,t}$. Finally, the real profits are

$$f_t^c = y_t^c - w_{P,t}^c n_{P,t}^{c,d} - w_{I,t}^c n_{I,t}^{c,d} - r_t^{k^c} u_t^{k^c} k_{t-1}^c.$$
(31)

Housing Sector The total production, as expressed by (18), must satisfy the aggregate demand for the sector

$$y_t^h = i_{P,t}^h + i_{I,t}^h, (32)$$

the real profits are

$$f_t^h = q_t^h y_t^h - w_{P,t}^h n_{P,t}^{h,d} - w_{I,t}^h n_{I,t}^{h,d} - r_t^{k^h} u_t^{k^h} k_{t-1}^h - r_t^l l_{t-1},$$
(33)

and the total housing stock is $h_t = h_{P,t} + h_{I,t}$.

Labour Input The nominal wage rigidity induces a loss in the number of hours worked supplied due to nominal wage dispersions. Schmitt-Grohe and Uribe (2007) show that these price dispersions can be expressed as

$$s_{i,t}^{j} = \left(1 - \xi^{w^{j}}\right) \left(\frac{\tilde{w}_{i,t}^{j}}{w_{i,t}^{j}}\right)^{-\theta^{n^{j}}} + \xi^{w^{j}} \left(\frac{\left(\pi^{c}\right)^{\iota^{w^{j}}} \left(\pi_{t-1}^{c}\right)^{1-\iota^{w^{j}}}}{\pi_{t}^{c}}\right)^{-\theta^{n^{j}}} \left(\frac{w_{i,t-1}^{j}}{w_{i,t}^{j}}\right)^{-\theta^{n^{j}}} s_{i,t-1}^{j},$$
(34)

and the labour supply-demand relation is given by $n_{i,t}^j = n_{i,t}^{j,d} s_{i,t}^j$.

Aggregate Economy The real GDP is therefore given by

$$y_t = y_t^c + q_t^h y_t^h. aga{35}$$

3 Empirical Strategy

In order to evaluate the performance of the model, we use a combination of calibrated and estimated parameters. Our choice to calibrate some of the parameters was mainly based on the lack of data for some of the model variables, particularly those describing the production functions and the wealth and income distributions. This section first describes our calibration approach, then presents the details regarding the estimation procedure, and concludes with a presentation of the data.

3.1 Calibration

The model is calibrated on a quarterly basis. Table 1 summarizes our calibration, while Table 2 displays the steady states of the model and observed values of corresponding data. We calibrated this set of parameters because they are either difficult to estimate given the information contained in the model²⁶ or they are better identified using other information. For instance, some parameters are set to achieve target values for steady states while others are set to commonly used values in the literature.

We set the steady state annual inflation rate at 2 percent, this value being the target of the inflation-control policy implemented by the Bank of Canada. The steady states of nominal and real interest rates reflect the lender's degree of time preference, β_P , and the steady-state gross inflation rate. We use an annual real rate of return of 2.77 percent (the average over our sample), which yields a β_P of 0.9925. As for the calibration of the borrower's time discount factor, based on the sample mean of R_t^m , we choose a value of 0.9844, which is in range of other studies that estimated or calibrated this parameter (Krusell and Smith, 1998; Iacoviello, 2005; Iacoviello and Neri, 2010; Gelain et al., 2013) and translates into a desire for borrowing. We are departing here from a common strategy used in previous studies estimating models of housing dynamics that assumes zero steady-state inflation. Given that, up to the first order, the steady state represents the unconditional mean of the variables, our approach has the advantage of centring the model closer to the unconditional mean in the data. The inflation expectations of 2 percent being well anchored in Canada by economic agents, our approach then has the desirable property to disentangle the expectations on components of nominal interest rates, which are the real interest rate

²⁶For identification testing, we compute the Fisher information matrix. It is a property of the model itself, and is independent of any data. It represents the maximum amount of information we can find in the data assuming the data are really generated by the model data-generating process (DGP) for a given parametrization. In that sense, it is a local identification test. We compute two approaches: a time-domain and a frequency-domain, and use a singular value decomposition to learn more about which parameters (or combinations of them) are identified the best or the worst. In our case, the intratemporal elasticity of substitution between sectoral labour supplies, the depreciation rates, the intratemporal elasticity of substitution across different varieties of intermediate goods and the wage-elasticities of demand has been revealed to be not or weakly identifiable.

and inflation rate.

The share of patient households (α) in the model is 0.25, representing the top quartile of households in the model economy. Parameter α determines the labour share of income and, indirectly, the physical capital wealth²⁷ and the distribution of real estate wealth. By targeting the top quartile of households and setting α at 0.25, we are able to target some important ratios: (i) the patient households own 72 percent of total wealth, which is broadly in line with financial data for that quartile;²⁸ (ii) the patient households earn 75 percent of total labour income, which is in line with income data;²⁹ and (iii) mortgage debt as a share of GDP is 0.89 (in the top of the distribution in our sample). We are departing here from the commonly used value in the literature, estimated at 0.79 by Iacoviello and Neri (2010) based on macroeconomic data and used by Lambertini, Mendicino, and Punzi (2010) and Lambertini et al. (2013b). We conduct local identification tests and find that it is not possible to identify this parameter in the absence of wealth data.³⁰

Following Iacoviello and Neri (2010), the quarterly depreciation rates for housing and capital in both the consumption and housing sectors are set at 0.01, 0.025 and 0.03, respectively, implying annual depreciation rates of 4.06 percent, 10.38 percent and 12.55 percent, respectively. Likewise, the prices and wages markups θ^c and θ^{n^j} are set at 7.67, which yields steady-state markups of 15 percent for intermediate goods producers and households. All those values are common in the literature.

The capital share of income in the consumption sector, γ^c , is set at 0.25. In the housing sector, we set the capital and land share of income γ^h and γ^l at 0.10 and 0.35, respectively. These factor shares, along with a weight of housing service in the utility function ϵ_{ss}^h at 0.75, the intertemporal elasticities $\sigma_P^x = \sigma_I^x = 1.2$, following Dorich, Johnston, Mendes, Murchison, and Zhang (2013), the intratemporal elasticities $\theta_P^x =$

²⁷Patient households own all the physical capital wealth.

²⁸See the Survey of Financial Security from Statistics Canada.

²⁹See the World Top Income Database available on Emmanuel Saez's website.

³⁰Estimation of the model with α as an estimated parameter yields a bi-modal posterior distribution of α , with modes around 0.25 and 0.75. Without wealth data based on micro-studies and surveys, it is not possible to discriminate between those values.

 $\theta_I^x = 0.4$ and the depreciation rates, imply steady-state ratios of consumption, nonhousing investment and housing investment to real GDP of approximately 74, 15 and 8 percent, respectively, which are in line with the data in our sample. Moreover, these calibration choices imply ratios of business capital and housing wealth (together with α) to annual GDP of around 1.6 and 2.0, respectively. While the former is in line with data, the latter is 50 percent higher than the mean over our sample. The model is not disaggregated enough to allow the ratio of housing-investment-to-GDP and of housing-stock-to-GDP to match the data at the same time. We then focus on the flow and leave the stock unmatched. Finally, along with the estimated parameters, the land share of income implies that the value of residential land is around 80 percent of GDP, a value close to the empirical data. The parameters η_P and η_I are set at 2.5 and 1.5, respectively, and the steady-state ϵ_{ss}^n is set so the steady-state labour supply by borrowers is 20 percent higher than by the lenders. The intratemporal elasticity of substitution between sectoral labour supplies is set at 10 for both households, yielding a share of total hours worked of 0.92 in the consumption sector, a value close to the data.

Finally, the LTV ratio is set at 0.91, which is the average value in Canada over the last few decades, while $\epsilon_{ss}^{R^m}$ is set at 0.066 to match the average quarterly spread between the risk-free and the 5-year mortgage rates over the last 30 years.

3.2 Bayesian Approach

The noncalibrated parameters are estimated by using a Bayesian approach (see DeJong, Ingram, and Whiteman (2000); Lubik and Schorfheide (2006); An and Schorfheide (2007)). In order to compute the likelihood for a given set of parameters, we solve a log-linear approximation of the equilibrium conditions in the neighborhood of the non-stochastic steady-state (Blanchard and Kahn, 1980; Klein, 2000; Sims, 2002).³¹ The solution takes the form of a state-space model that is used to compute the likelihood

 $^{^{31}{\}rm We}$ use a modified version of the Klein (2000) algorithm available in the IRIS Toolbox (http://iristoolbox.codeplex.com/).

function, and, given the linear solution and the assumption of normally distributed shocks, the Kalman filter is used to evaluate the likelihood.

Given the likelihood function, we characterize the posterior distribution in two steps. First, we transform the data into a form suitable for computing the likelihood function, we use the parameter and the model prior distributions to incorporate additional information into the estimation, and we maximize the posterior using numerical methods.³² Finally, we use a metropolis posterior simulator to evaluate the behaviour of the posterior distribution and to draw model parameters from the posterior distribution.³³

Parameter's Prior Distributions The advantage of using priors is to take our *a priori* beliefs into account in estimating the parameters of the model. The choice of priors is described in the second, third and fourth columns of Table 3 and Table 4 for the noncalibrated parameters, and Table 5 for the measurement errors. The prior distributions are guided by the constraints in these parameters and are either consistent with previous studies (Levin, Onatski, Williams, and Williams, 2006; Del Negro, Schorfheide, Smets, and Wouters, 2007; Justiniano et al., 2010; Iacoviello and Neri, 2010; Schmitt-Grohe and Uribe, 2012) or fairly diffuse and relatively uninformative.

To reflect their strict positivity, we set a Gamma prior on the investment adjustment costs (ϕ^{k^c} and ϕ^{k^h}) around 5 with a standard error of 2. We select a Beta prior for the Calvo price and wage parameters (ξ^{p^c} , ξ^{w^c} and ξ^{w^h}) and the inflation indexation parameters (ι^{p^c} , ι^{w^c} and ι^{w^h}), as they belong to the interval [0, 1), and due to a lack of consensus on their values in the literature (Christiano, Eichenbaum, and Evans, 2005; Smets and Wouters, 2007), we set the prior mean at 0.5, with a standard deviation of 0.22.

For all the persistence parameters governing the exogenous processes, we use a Beta prior with a mean equal to 0.80 and a standard deviation equal to $0.1.^{34}$ For all

 $^{^{32}}$ We use the Active-set algorithm in KNITRO.

³³We use the mode obtained in the first step as a starting point and an adaptive random-walk Metropolis posterior simulator with 500,000 draws with 100,000 burn-in draws and target acceptance ratio of 0.234.

 $^{^{34}}$ A Beta prior with a mean equal to 0.5 and a standard deviation equal to 0.22 yields the same

innovation and measurement error standard deviations, we use an inverse-Gamma prior with a mean equal to 0.1 and a standard deviation equal to 0.2. These priors are quite disperse and were chosen to generate volatility in the endogenous variables that is broadly in line with the data. Their covariance matrix is assumed to be diagonal.

For the monetary policy specification, we base our priors on a standard Taylor rule with interest rate smoothing that responds gradually to inflation and output gap. We use a Beta prior for ρ_r and ρ_y with means of 0.8 and 0.1 and standard deviations of 0.3 and 0.025, respectively, and a Gamma prior with mean of 3.5 and a standard deviation of 0.5 for ρ_{π^c} . These priors are in line with previous Canadian studies (Christensen et al., 2009; Dorich et al., 2013).

Model's Prior Distributions We also use additional information to estimate the model, by implementing a novel approach using model priors (Andrle and Benes, 2013). In contrast with parameter priors, model priors are those about the model's features and behaviour as a system, such as the covariance and correlation. While being consistent and reasonable at the parameter level, parameter priors can result in unreasonable aggregate model properties, different from the researcher's beliefs, due to the nonlinear mapping of parameters into the model's properties. In contrast, a prior about system properties creates direct stochastic restrictions on the combinations of parameters. In our case, since we focus on housing-market-related business cycles, spillovers from housing wealth to consumption level and the notion of boom-bust, we select correlation to be the most relevant model priors to implement. More specifically, we use the first- to third-order cross-correlation between consumption, residential investment, non-residential investment, house price and mortgage debt, and we apply a normal prior with the mean being the sample first- to third-order cross-correlation computed on the filtered data used in the estimation (see next section on Data), and standard deviation set at 0.1. The choice of priors is described in Table 6.

estimation results on these parameters.

Data To estimate the model, we use Canadian quarterly data for the period 1983Q3 to 2014Q4. The vector of observables used for the estimation includes 15 variables: real consumption, residential investment, non-residential investment, and mortgage debt per capita; real house and capital prices; nominal short- and long-term interest rates; core CPI inflation rate; and finally hours worked per capita, real wage and capacity utilization rate in both consumption and housing sectors. In order to remove the trend and isolate the cyclical component, we apply the one-sided Hodrick-Prescott (HP) filter on most of data.³⁵ The interest rates are detrended by removing a linear time trend, and the core CPI inflation rate and the capacity utilization rates are demeaned. Figure 1 plots the time series. A detailed description of the series used in the estimation is provided in Appendix B. In addition, we include i.i.d. measurement errors for hours worked, wages and capacity utilization rate for both sectors.

Our set of observables includes more variables than most previous DSGE estimations for housing market dynamic models. We consider series that are of general interest for policy analysis usually used in the literature, such as consumption, investments, wages, hours worked, inflation and interest rates. Our data set also includes variables that may *a priori* help us identify several features of the model. For instance, sectoral data, such as hours worked, wages and capacity utilization rates, will be useful in characterizing movements and correlation that are sector-specific and could be hidden in aggregated data. As an example, hours worked and real wages in the housing sector experience around two times the volatility of their counterparts in the consumption sector (see Figure 1). Since the consumption sector accounts for 75 percent of the model-consistent total GDP, using aggregated data would hide this sectoral volatility. We construct the sectoral data by linking the model definitions to their Standard Industry Classification (SIC) and North American Industry Classification System (NAICS) counterparts. The housing sector is approximated by the construction sec-

³⁵The model solution takes the form of a backward-looking state-space system, and a non-causal two-sided HP filter would contradict this structure. The better option is to use the backward-looking one-sided HP filter (Stock and Watson, 1999). We adjust for seasonality before any computation when necessary.

tor.³⁶ The construction sector also includes the non-residential construction, but since we focus on the cyclical component and not the level, we make the conjecture that the cycles in both non-residential and residential construction are approximatively similar. The consumption sector includes the rest of goods- and services-producing industries, excluding all government components. Finally, mortgage debt contains information on the reallocation of debt between agents and the preference on consumption and housing services.

Monetary Policy Since the Bank of Canada introduced its inflation-targeting regime in 1991, the Taylor-type monetary policy rule appropriately describes the behaviour of the short-term interest rate path from 1992 to 2014, but is less likely to reflect its behaviour from 1983 to 1991. Therefore, we estimate the model in two steps. First, we use a restricted sample covering 1992Q1 to 2014Q4 without the time-varying inflation target process to recover the parameters of the monetary policy rule. Second, in the full sample estimation covering 1983Q3 to 2014Q4, the parameters of the monetary policy rule are calibrated and time-varying inflation targeting is introduced. This two-step process allows us to both impose the same monetary policy rule over the full sample and avoid introducing a bias on estimated parameters by estimating the model on the sample when the assumed functional form of the monetary policy rule is likely to apply.

4 Empirical Results

In this section, we first describe the estimated posterior distribution, paying attention to the parameters describing the housing market dynamics. We then perform a posterior analysis to establish the extent to which the model can fit the data.

 $^{^{36}\}mathrm{Codes}\;\mathrm{F}$ in SIC and 23 in NAICS.

4.1 Posterior Distributions

The estimated posterior distributions of the noncalibrated parameters are summarized in Table 3 and Table 4, the measurement errors are presented in Table 5, while the model's properties are shown in Table 6. In general terms, the information contained in the likelihood significantly updates the assumed priors for all the parameters, given the marked differences in the statistics describing these two distributions.

The capital adjustment costs seem to differ across sectors. These results could imply that the model requires partial capital mobility across sectors in order to better approximate the data. Cumulated with the imperfect mobility in the labour market, this means that the real frictions caused by imperfect mobility play a significant role in the suboptimal allocation of resources relative to the perfect mobility scenario.

With autoregressive parameters being, in general, higher than 0.85, the estimated exogenous processes are generally persistent, except for the technology process in the consumption sector and the investment-specific process, with parameters equal to 0.48 and 0.68, respectively. Labour supply is the most persistent process, with an autoregressive parameter equal to 0.9991. In terms of volatility, among the estimated standard error of the exogenous processes, the interest rate spread shock seems to be the most volatile, followed by the investment-specific shock. However, we will see in the next section that the latter shock is not a main driver of the forecast error variance decomposition, mainly because of its persistence.

Regarding parameters measuring nominal rigidities, the estimate of θ^{p^c} (0.62) implies that prices are reoptimized frequently, once every 2 quarters. However, given the positive value of the indexation parameter ($\iota^{p^c} = 0.96$), prices change every period at a rate mostly equal to the Bank of Canada target inflation rate, and therefore not optimally in response to a change in nominal costs. As for wages, we find that stickiness in the housing sector ($\theta^{w^h} = 0.97$) and the consumption sector ($\theta^{w^c} = 0.98$) are almost equal. While being reoptimized infrequently, once every 30–40 quarters, wages are indexed every period to compensate the steady-state inflation ($\iota^{w^h} = 0.84$ and $\iota^{w^c} = 0.37$) and the past inflation.

Finally, with a large weight on inflation ($\rho_{\pi^c} = 4.23$) and a fairly small weight on output gap ($\rho_y = 0.30$), estimates of the parameters of the monetary policy rule are in line with previous evidence (Christensen et al., 2009; Dorich et al., 2013) and the parameters used at the Bank of Canada. In terms of the three monetary disturbances, the shock to interest rate spread is the most volatile, and more persistent than the shock to inflation targeting. The monetary policy shock standard error is perfectly in line with previous studies with Canadian data (Christensen et al., 2009; Dorich et al., 2013).

4.2 Second Moments

Table 7 presents first-, second- and fourth-order auto- and cross-correlations for a set of selected model variables for both the data moments and asymptotic (modelbased) moments evaluated using the posterior mode. While the model underestimates the first- and fourth-order autocorrelations of mortgage debt and the fourth-order autocorrelation of residential and non-residential investment, it is able to replicate well all the other autocorrelations of the data, with the theoretical autocorrelation always being closed from its data counterpart. The model also matches both the sign and the level of the cross-correlations for most of the desired relationship being studied. The theoretical zeroth- to fourth-order cross-correlations of consumption with housing investment and house price are all in line, both in sign and level, with their data counterparts, which is an important feature to match to attain the desired volatility and co-movement in boom-bust scenarios. Lastly, the model-based correlation between consumption and mortgage debt is also of the expected sign, but is twice the level of correlation seen in the data. Overall, the model seems to properly replicate the behaviour of the observables.

4.3 Variance Decomposition

After establishing the extent to which the estimated model can replicate the business cycle observations, we proceed with the variance decomposition. In this section we discuss both the historical and forecast error variance decomposition.

Historical Figure 3 presents the historical variance decomposition for a set of selected observables for the period 1983Q2 to 2014Q4. We present results by grouping shocks into five classes: (i) monetary policy and financial, which includes $\varepsilon_t^{\epsilon^{\chi}}$, $\varepsilon_t^{\epsilon^{\pi^c}}$, $\varepsilon_t^{\epsilon^R}$ and $\varepsilon_t^{\epsilon^{R^m}}$; (ii) demand, which is $\varepsilon_t^{\epsilon^b}$; (iii) supply, which includes $\varepsilon_t^{\epsilon^n}$, $\varepsilon_t^{z^c}$, $\varepsilon_t^{z^h}$ and $\varepsilon_t^{z^{i^k}}$; (iv) housing, which is all housing demand shocks (surprise and news); and finally (v) measurement errors.

Focusing on the decomposition of the housing investment and house price first, it is clear that their short-run variability is mostly accounted for by the housing demand shocks and the monetary policy and financial shocks. Indeed, both types of shocks seem to have driven the decline in both supply and demand sides in the two last recessions of 1991 and 2008, while the boom in the 1980s was driven by only the latter.

As expected, most of the fluctuations in non-residential investment in both production sectors are mainly driven by the supply-specific shocks. However, a non-negligible share is also explained by monetary policy and financial shocks, which affect the intertemporal reallocation of resources of lenders over time, and then have a direct impact on investment decision.

Forecast Error We now proceed with the forecast error variance decomposition. Table 8 presents the unconditional variance decomposition of the observables for all the shocks in the model, the last column being the sum of the contributions of all measurement errors.

In terms of explaining the consumption fluctuations, the labour supply, the credit,

the housing demand (both news and surprise) and the inflation-targeting shocks appear to be the most significant. They explain respectively 54, 18, 10.3 and 9.8 percent of the consumption volatility. The labour supply shock directly affects the labour income of the agents and therefore all economic decisions made by households, while the inflation-targeting shock causes, for a given increase in price level, variation in the reaction of the monetary policy across periods. Finally, the credit shock affects the impatient households' capacity to borrow against their collateral, therefore affecting their consumption. The other drivers do not seem to play a significant role in explaining consumption fluctuations. These results are in contrast with other studies that identified technology shock in the consumption sector and monetary policy shock among the main drivers of consumption volatility.

The same analysis applies to non-residential investment and capital price, while replacing the credit shock with the investment-specific shock as the third driver. Since these variables are driven by decisions of patient households and are not affected directly by the credit shock (but only by spillovers from impatient households' reactions to credit shock), these results were expected. Finally, house price and residential investment forecast error variances are mostly driven by labour supply, housing demand and total factor productivity (TFP) in housing sector shocks, as expected.

4.4 Shock Responses

In the last section, we analyzed how well the estimated model can replicate the business cycle observations. It is also important to understand the dynamics in the model implied by the shocks. In this section, we focus on five of them that explain most of the volatility in the model. All results presented in this section are the model's responses to a one-standard-deviation shock.

Housing Demand Figure 4 plots impulse responses to the estimated housing demand shocks, both surprise and news. Overall, it raises on impact house price and returns on housing investment. Since borrowers' collateral is linked to house price, they
can increase their level of borrowing and consumption. Given their higher marginal propensity to consume, the effects on total consumption are positive and entirely driven by borrowers, their consumption increase being high enough to compensate the decrease in lenders' consumption.

Housing demand shocks generate the co-movement between house price, total consumption, residential investment and hours worked (not shown) in both sectors of production observed in the data, especially during periods of housing booms. Shocks affect economic choices and, in particular, the housing investment and credit decisions of households. The occurrence of a positive housing demand shock prompts appreciation in housing price and fuels current housing demand. Consequently, housing investment rises quickly on impact, with a peak increase of more than 1.7 percent. House price follows the same curve, with a peak increase of 1.4 percent on impact. Mortgage debt increases significantly, by close to 15 percent, reflecting both the increase in housing investment and the increase in house price that affect the value of all the undepreciated housing stock. This increase in the collateral value boosts the consumption of borrowers and causes inflationary pressures, which has the effect of increasing interest rates, both short- and long-term. Moreover, due to limits to credit, borrowers increase their labour supply in order to raise funds for housing investments. However, coupled with a decrease in non-residential investment, wages rise. The increase in consumption and housing investment also makes GDP rise. Thus, housing demand shocks in this model generate procyclicality among relevant variables.

The story behind news shocks is similar, but the dynamics are different, since news shocks generate reactions only via the expectation channel. When news about higher future housing demand arrives, it is optimal for agents to immediately start increasing their housing stock to take advantage of the capital gain. The marginal gain is higher for impatient than patient households, as it increases the marginal utility of housing stock for both but also relaxes the borrowing constraint for the former, allowing them to borrow and consume more. Patient households sell their houses to impatient households. Optimistic expectations lead to excessive housing investment, thereby causing a boom in the housing market not based on fundamentals.

Labour Supply Figure 5 presents the model's response to a shock in labour supply. This shock induces a greater disutility of hours worked to agents, causing an immediate decrease in hours worked in both sectors. This decrease leads to an increase in real wages in both sectors as the productivity increases slightly. The marginal cost of production is driven up and gradually transmitted to the inflation in the consumption sector, which drives interest rates up via the monetary policy response. The decrease in labour income and the increase in borrowing costs lead to a decrease in housing investment from the borrowers and a real house price decline, thereby reducing collateral values. Overall, all economic aggregates react negatively in response to negative income shock.

Monetary and Inflation Targeting Figure 6 plots the effects of monetary policy shocks. The temporary shock leads to a rise in the nominal and real short-term interest rates, and a fall in output, consumption and residential and non-residential investment. In line with the stylized facts on monetary policy shocks, real wages fall (not shown). The largest effect on consumption is about 1.5 times that on non-residential investment. Overall, these effects are consistent with the evidence found in the literature.

Finally, Figure 6 presents the model's response following a shock in inflation targeting. An increase in the inflation target means that, following an increase in inflation, the central bank will not increase the interest rate as much as in steady state. The effects of a persistent change in the inflation objective are strikingly different from the monetary policy shock in one aspect. First, there is a liquidity effect, as nominal interest rates start increasing immediately as a result of the increased inflation expectations. Inflation picks up immediately, driven by an increase in consumption and housing investment. Interest rates continue to increase in response to higher inflation, this time at a slower pace. **Interest Rate Spread** Figure 7 plots the effects of interest rate spread shocks. The temporary shock leads to a rise in the nominal and real long-term interest rates, and a fall in output, consumption and residential and non-residential investment. In line with the stylized facts on monetary policy shocks, real wages fall (not shown). The real short-term interest rate declines to respond to negative outlook on inflation. Overall, these effects are consistent with the evidence found in the literature.

LTV Finally, Figure 8 plots impulse responses to the estimated LTV shocks, which are credit shocks. A positive shock on LTV relaxes the borrowing constraint of impatient households, allowing them to borrow and consume more. However, to finance this borrowing, patient households reduce their consumption and housing investment, inducing a house price decline, and constraining the borrowing constraint of impatient households. Overall, consumption and non-residential investment increase, but the effect on consumption is short-lived.

4.5 Robustness

As mentioned in Section 3.2, we estimate the model in two steps. While it is not the primary purpose of this procedure, this two-step estimation provides robustness analysis by estimating our model over a subsample. Overall, the results described for the full sample remain true when considering the subsample, with a notable exception of the persistence of the LTV exogenous process, which is less persistent by a factor of 2 than in its full sample counterpart. Because of similarity of results, we do not reproduce the robustness analysis in this paper, but all results are available upon request.

5 Housing Market News-Driven Boom-Bust Cycles and Loanto-Value Ratio

Since we now have an estimated model able to replicate business cycle properties of the housing market in the Canadian economy, in this section, we first highlight key findings regarding the transmission mechanism of news shocks and describe the role of news shocks in housing market dynamics. Through simulation, we analyze the impacts of news shocks on selected variables over the business cycle and show that news shocks can generate boom-bust housing market cycles. Then we examine the effectiveness of implementing different countercyclical LTV ratios and compare it to the effectiveness of a simple Taylor-type monetary policy rule augmented with house price inflation.

5.1 News Shocks and Housing Market News-Driven Boom-Bust Cycles

Our paper contributes to the news-driven business cycle literature. It is well-known in the literature that rational-expectations DSGE models can hardly generate newsdriven boom-bust cycles. In the real-business cycle literature, positive news about future TFP shocks create a strong wealth effect, inducing a decline in hours worked. Agents then produce less and invest less to finance their consumption increase. Therefore, positive news shocks create a recession. The economic outcome becomes even worse when the positive news shocks are unrealized. There is a stream of literature that studies which mechanisms could potentially generate news-driven business cycles (see, among others, Beaudry and Portier (2004) and Jaimovich and Rebelo (2009)).

In general, macroeconomic models of housing markets rely on mainly fundamental developments in the economy to explain fluctuations in house price and residential investment (notable exceptions are Kanik and Xiao (2014) and Lambertini et al. (2013b)). However, survey evidence shows that house price dynamics are greatly related to macroeconomic expectations, especially to optimism about future house price appreciations (Lambertini, Mendicino, and Punzi, 2013a). Also, real housing price can significantly deviate from economic fundamentals, as during the 1998–2007 housing boom episode (Shiller, 2007).

In this paper, we generate news-driven business cycles and co-movement among economic aggregates by relying on two features of the model: (i) the existence of heterogeneous agents; and (ii) the collateral channel. When news about higher future housing demand arrives, it is optimal for agents to start immediately increasing their housing stock to take advantage of the capital gain. The marginal gain is higher for impatient than patient households, as it increases the marginal utility of housing stock for both but also relaxes the borrowing constraint for the former, allowing them to borrow and consume more. Patient households sell their houses to impatient households. Optimistic expectations lead to excessive housing investment, thereby causing a boom in the housing market not based on fundamentals. Once the news shocks are found to be unrealized, buyers revert their actions and a bust in the housing market follows. Therefore, our model allows us to study the effect of macroprudential policy in a context where agents can have (over-)optimistic expectations about the future and react by contracting more debt under fixed long-term contracts.

Figure 9 presents the simulated impact of anticipated shocks on housing demand for key macroeconomic variables. Three simulations are presented in the figure. The **unanticipated shocks** case presents a scenario consisting of a series of four unanticipated shocks (i.e. $\varepsilon^{\epsilon^{h},0}$) on housing demand. We assume an increasing value of $\varepsilon^{\epsilon^{h},0}$ from t + 5 to t + 8: (i) $\varepsilon^{\epsilon^{h},0}_{t+5} = 0.5\sigma_{\epsilon^{h},0}$; (ii) $\varepsilon^{\epsilon^{h},0}_{t+6} = 1.0\sigma_{\epsilon^{h},0}$; (iii) $\varepsilon^{\epsilon^{h},0}_{t+7} = 1.5\sigma_{\epsilon^{h},0}$; and (iv) $\varepsilon^{\epsilon^{h,0}}_{t+8} = 2.0\sigma_{\epsilon^{h},0}$. The **anticipated shocks** case presents a scenario consisting of a series of four anticipated shocks learned in t but unrealized (i.e. revised by an equivalent negative unanticipated shock in the period it was supposed to happen): (i) $\varepsilon^{\epsilon^{h},4}_{t+1} = 0.5\sigma_{\epsilon^{h},4}$ but revised with $\varepsilon^{\epsilon^{h},0}_{t+5} = -0.5\sigma_{\epsilon^{h},4}$; (ii) $\varepsilon^{\epsilon^{h},4}_{t+2} = 1.0\sigma_{\epsilon^{h},4}$ but revised with $\varepsilon^{\epsilon^{h},0}_{t+6} = -1.0\sigma_{\epsilon^{h},4}$; (iii) $\varepsilon^{\epsilon^{h},4}_{t+3} = 1.5\sigma_{\epsilon^{h},4}$ but revised with $\varepsilon^{\epsilon^{h},0}_{t+6} = -1.5\sigma_{\epsilon^{h},4}$; and (iv) $\varepsilon^{\epsilon^{h},4}_{t+6} = 2.0\sigma_{\epsilon^{h},4}$ but revised with $\varepsilon^{\epsilon^{h},0}_{t+8} = -2.0\sigma_{\epsilon^{h},4}$.

As expected, the unanticipated shocks plotted in Figure 9 start to have an impact in t + 5, when the shock happens. Therefore, the impacts under this scenario are the same as the impulse response following a housing demand shock described in Section 4. We observe a positive co-movement between housing investment, house price and consumption, but also a monetary policy reaction following the slight increase in inflation and the deviation of GDP from its steady-state value. Finally, housing demand increases following the four positive shocks from t + 5 to t + 8.

The story is different for the anticipated shocks scenario. News shocks generate the co-movement between house price, consumption and residential investment, but also hours worked (not shown) in both sectors of production observed in the data, especially during periods of housing booms. News shocks affect economic choices and, in particular, the housing investment and credit decisions of households differently than unanticipated shocks. Expectations about the occurrence of positive housing demand shocks immediately generate beliefs of future appreciation in housing price and fuel current housing demand. All the agents start to learn about the positive news shocks at the same moment in t + 1. Consequently, housing investment rises quickly, with a peak increase of near 6 percent in t + 5. House price follows the same curve, with a peak increase of 4 percent in t+5. Mortgage debt increases significantly, by more than 20 percent, reflecting the increase in housing investment, but also the increase in house price that affects the value of all the undepreciated housing stock. This increase in the collateral value boosts the consumption of borrowers and fuels inflation, inducing a rise in interest rates. Overall, as news shocks spread, the value of housing collateral increases and the rise in house price is, thus, coupled with an expansion in household credit and consumption. Moreover, due to limits to credit, borrowers increase their labour supply in order to raise internal funds for housing investments. For the decrease in non-residential investment to be coupled with an increase in hours, wages rise. The increase in consumption and housing investment also causes GDP to rise. Thus, news shocks in this model generate procyclicality among relevant variables. However, in t + 5, agents learn about the housing demand and revise their views on the current state of the economy: positive housing demand shock does not occur. Housing investment and house price start to decline on impact, followed by mortgage debt. The collateral value then starts to drop and agents have to revise their consumption level. The same mechanism occurs every time when the positive housing demand news shock does not materialize. From t+5 to t+9, housing investment declines by 6 percent and house price by 4 percent. Moreover, from peak to trough, consumption level declines by nearly 15 percent, and the real GDP declines

by close to 10 percent, generating a recession. All of this resulted from unrealized expectations.

Overall, the two case scenarios suggest that news shocks could play an important role in boom-bust housing market cycles, as they can generate co-movement between consumption, housing investment and house price, similar to what is observed in the data, especially during periods of housing booms. However, the co-movement is not the same as that expected for non-residential investment. In a closed-economy model (like the one we study), it would be hard to generate the right co-movement with non-residential investment, as the increase in marginal utility and decrease in marginal cost from housing stock would drain the resources in the economy, therefore reducing investment and capital stock. However, in an open-economy model, where agents can finance their capital stock with foreign saving, we would expect to attain the right co-movement for non-residential investment as well.

5.2 Countercyclical LTV

We now study the effectiveness of implementing a countercyclical LTV ratio to reduce or eliminate the amplitude of the boom-bust cycles described above (i.e. **anticipated shocks** case). First, we consider two countercyclical LTV ratios. In both cases, we assume that the central bank continues to follow the estimated monetary policy rule and we allow the LTV ratio to vary around its long-run setting of 91 percent. The first rule considered is based on the deviation of house price from its steady state

$$\omega_t = \omega \left(\frac{q_t^h}{q^h}\right)^{-\phi^\omega},$$

while the second rule is based on the deviation of the debt-to-GDP ratio from its steady state

$$\omega_t = \omega \left(\frac{M_t/y_t}{M/y}\right)^{-\phi^{\omega}},$$

with ϕ^{ω} being the countercyclical parameter.³⁷ Finally, we compare the results of these regulatory LTV policies with the performance of a Taylor-type monetary policy rule augmented with house price inflation

$$R_{t} = \rho_{r}R_{t-1} + (1 - \rho_{r})\left(R + \rho_{\pi^{c}}\left(\frac{\prod_{v=1}^{4}\pi_{t+v}^{c}}{4} - \epsilon_{t}^{\pi^{c}}\right) + \rho_{y}\left(Y_{t} - Y\right) + \rho_{\pi^{h}}\left(\pi_{t}^{h} - \pi^{h}\right)\right)$$

Figure 10, Figure 11 and Figure 12 show the simulation results. We implemented three cases of macroprudential policy. For those with countercyclical LTV, we implemented the rules with $\phi^{\omega} = \{0.0, 0.5, 1.0\}$, with the countercyclical parameter equal to 0, replicating the **anticipated shocks** case (no countercyclical LTV) to facilitate comparisons. In the case of monetary policy, three values of the parameters are considered, namely $\rho_{\pi^h} = \{0.0, 1.0, 2.0\}$, with the house price inflation parameter equal to 0, replicating **anticipated shocks** case (no countercyclical LTV) to facilitate comparisons. When policy is based on house price, for both parameter values the countercyclical LTV ratio does not reduce the surge in housing investment and house price. Expectations about the occurrence of positive housing demand shocks still immediately generate beliefs of future appreciation in housing price and fuel current housing demand. However, the transmission mechanism creating a spillover effect on consumption via loosening of the collateral constraint is greatly reduced when $\phi^{\omega} = 0.5$ and eliminated when $\phi^{\omega} = 1.0$. Therefore, we still experience a house price correction of nearly 6 percent and a housing investment decrease of 8 percent when agents realize that the expectations do not materialize, but this does not lead to a recession. When policy is based on the debt-to-GDP ratio, as in the previous rule studied, house price and housing investment are not affected by LTV, because the housing demand news shocks dominate, but the transmission mechanism is greatly reduced. However, when $\phi^{\omega} = 1.0$, the wealth effect via the collateral constraint is still materialized and we still observe a decline in consumption.

³⁷This choice of tested policies is based on the literature but remains arbitrary. Many other rules for countercyclical LTV could be tested, but a full comparative study based on welfare criterion would be necessary. This paper can serve as a starting point to welfare studies.

Finally, we consider a modified Taylor-type rule augmented with house price inflation. As expected, the effects of including the house price inflation in the monetary policy rule are more diffuse and affect all the macroeconomic variables. It helps to reduce the house price and housing investment impacts of news shocks on housing demand; however, it is less effective than a sectoral policy like the LTV ratio to uniquely target the spillover of the wealth effect on consumption via the loosening of the collateral constraint.

6 Conclusion

In this paper, we perform an analysis to determine how well the introduction of a countercyclical LTV ratio can reduce household indebtedness and housing price fluctuations compared with a monetary policy rule augmented with house price inflation. To this end, we construct a New Keynesian model in which a fraction of households borrow against the value of their houses and we introduce news shocks on housing demand. We estimate the model with Canadian data using Bayesian methods.

We find that the introduction of news shocks can generate a housing market boombust cycle, the bust following unrealized expectations on housing demand. Housing values affect agents' net worth and their ability to borrow and spend. A housing cycle can therefore trigger co-movements in aggregate economic activities and generate a boom-bust. When optimistic economic news is unrealized, the fluctuations that were not based on fundamentals are cancelled by the realization of the real shocks.

Our study also suggests that a countercyclical LTV ratio, especially the rule based on house price deviation from its steady state, is a useful policy to reduce the spillover from housing market to consumption and to lean against news-driven boom-bust cycles in housing price and credit generated by expectations of future macroeconomic developments.

As pointed out in Iacoviello and Neri (2010), a good part of the fluctuations in housing price and housing investment observed in the data are viewed by the model as the outcome of the exogenous shift in housing demand. This shock potentially includes unmodelled features of the model. The housing investment is mainly made at the household level, while our data are per capita. With the constant decrease in the number of persons per household observed since the beginning the 1970s, this dynamic is probably captured within the housing demand shock. Also, using perturbation methods, it is hard to model exogenous change in policy, as the one we observed in regulatory LTV ratios over the last 15 years. Changes in LTV requirements could potentially have been captured in the housing demand shock. These elements are interesting questions for further research.

Given the results in this paper, we now have to enhance our understanding of housing boom-bust cycles and macroprudential policy in other respects. Our analysis is based on many assumptions. First, all data are perfectly measured without any vintage revisions. If data are revised, then the countercyclical LTV would not adequately measure the current state of the economy it is intended to target. Second, we supposed that the central authority managing the macroprudential policy can identify the news shocks from the unanticipated shocks. In the case of unanticipated shocks, the central authority does not necessarily want to react, since the housing demand increase is based on fundamentals. Third, we do not consider the possibility of precautionary savings, which can play a substantial role with the expectation channel. Finally, we perform simulation, while welfare analysis would consider more aspects, among them the reallocation of resources from patient to impatient households during the housing boom. We leave these for future research.

References

ADOLFSON, M., S. LASEEN, J. LINDE, AND M. VILLANI (2007): "Bayesian estimation of an open economy DSGE model with incomplete pass-through," *Journal of International Economics*, 72, 481–511.

ALPANDA, S., G. CATEAU, AND C. MEH (2014): "A policy model to analyze macro-

prudential regulations and monetary policy," Working Papers 14-6, Bank of Canada.

- ALPANDA, S. AND S. ZUBAIRY (2014): "Addressing household indebtedness: monetary, fiscal or macroprudential policy?" Working Papers 14-58, Bank of Canada.
- AN, S. AND F. SCHORFHEIDE (2007): "Bayesian analysis of DSGE models," *Econo*metric Reviews, 26, 113–172.
- ANDRLE, M. AND J. BENES (2013): "System priors: formulating priors about DSGE models' properties," IMF Working Papers 13/257, International Monetary Fund.
- BEAUDRY, P. AND F. PORTIER (2004): "An exploration into Pigou's theory of cycles," Journal of Monetary Economics, 51, 1183–1216.
- BLANCHARD, O. J. AND C. M. KAHN (1980): "The solution of linear difference models under rational expectations," *Econometrica*, 48, 1305–11.
- CALVO, G. A. (1983): "Staggered prices in a utility-maximizing framework," *Journal* of Monetary Economics, 12, 383–398.
- CAMPBELL, J. Y. (2013): "Mortgage market design," Review of Finance, 17, 1–33.
- CHRISTENSEN, I., P. CORRIGAN, C. MENDICINO, AND S.-I. NISHIYAMA (2009): "Consumption, housing collateral, and the Canadian business cycle," Working Papers 09-26, Bank of Canada.
- CHRISTIANO, L. J., M. EICHENBAUM, AND C. L. EVANS (2005): "Nominal rigidities and the dynamic effects of a shock to monetary policy," *Journal of Political Economy*, 113, 1–45.
- DEJONG, D. N., B. F. INGRAM, AND C. H. WHITEMAN (2000): "A Bayesian approach to dynamic macroeconomics," *Journal of Econometrics*, 98, 203–223.
- DEL NEGRO, M., F. SCHORFHEIDE, F. SMETS, AND R. WOUTERS (2007): "On the fit of new Keynesian models," *Journal of Business & Economic Statistics*, 25, 123–143.

- DORICH, J., M. K. JOHNSTON, R. R. MENDES, S. MURCHISON, AND Y. ZHANG (2013): "ToTEM II: an updated version of the Bank of Canada's quarterly projection model," Technical Reports 100, Bank of Canada.
- ERCEG, C. J., D. W. HENDERSON, AND A. T. LEVIN (2000): "Optimal monetary policy with staggered wage and price contracts," *Journal of Monetary Economics*, 46, 281–313.
- GELAIN, P., M. KOLASA, AND M. BRZOZA-BRZEZINA (2014): "Monetary and macroprudential policy with multi-period loans," 2014 Meeting Papers 575, Society for Economic Dynamics.
- GELAIN, P., K. J. LANSING, AND C. MENDICINO (2013): "House prices, credit growth, and excess volatility: implications for monetary and macroprudential policy," *International Journal of Central Banking*, 9, 219–276.
- HORVATH, M. (2000): "Sectoral shocks and aggregate fluctuations," *Journal of Mon*etary Economics, 45, 69–106.
- IACOVIELLO, M. (2005): "House prices, borrowing constraints, and monetary policy in the business cycle," *American Economic Review*, 95, 739–764.
- IACOVIELLO, M. AND S. NERI (2010): "Housing market spillovers: evidence from an estimated DSGE model," *American Economic Journal: Macroeconomics*, 2, 125–64.
- IMF (2012): "World Economic Outlook April 2012; Chapter 3: Dealing with household debt." Tech. rep., IMF, USA.
- (2013): "IMF Policy Paper: Key aspects of macroprudential policy," Tech. rep., IMF, USA.
- JAIMOVICH, N. AND S. REBELO (2009): "Can news about the future drive the business cycle?" *American Economic Review*, 99, 1097–1118.
- JUSTINIANO, A., G. E. PRIMICERI, AND A. TAMBALOTTI (2010): "Investment shocks and business cycles," *Journal of Monetary Economics*, 57, 132–145.

(2013): "Household leveraging and deleveraging," NBER Working Papers 18941, National Bureau of Economic Research, Inc.

- KANIK, B. AND W. XIAO (2014): "News, housing boom-bust cycles, and monetary policy," *International Journal of Central Banking*, 10, 249–298.
- KANNAN, P., P. RABANAL, AND A. M. SCOTT (2012): "Monetary and macroprudential policy rules in a model with house price booms," *The B.E. Journal of Macroeconomics*, 12, 1–44.
- KIYOTAKI, N. AND J. MOORE (1997): "Credit cycles," Journal of Political Economy, 105, 211–48.
- KLEIN, P. (2000): "Using the generalized Schur form to solve a multivariate linear rational expectations model," *Journal of Economic Dynamics and Control*, 24, 1405–1423.
- KRUSELL, P. AND A. A. SMITH (1998): "Income and wealth heterogeneity in the macroeconomy," *Journal of Political Economy*, 106, 867–896.
- LAMBERTINI, L., C. MENDICINO, AND M. T. PUNZI (2010): "Expectations-driven cycles in the housing market," Working Papers w201004, Banco de Portugal, Economics and Research Department.
- (2013a): "Expectation-driven cycles in the housing market: evidence from survey data," *Journal of Financial Stability*, 9, 518–529, re-examining the role of the state in the financial sector.
- LAMBERTINI, L., C. MENDICINO, AND M. TERESA PUNZI (2013b): "Leaning against boom-bust cycles in credit and housing prices," *Journal of Economic Dynamics and Control*, 37, 1500–1522.
- LEVIN, A. T., A. ONATSKI, J. WILLIAMS, AND N. M. WILLIAMS (2006): "Monetary policy under uncertainty in micro-founded macroeconometric models," in *NBER*

Macroeconomics Annual 2005, Volume 20, National Bureau of Economic Research, Inc, NBER Chapters, 229–312.

- LUBIK, T. AND F. SCHORFHEIDE (2006): "A Bayesian look at the new open economy macroeconomics," in NBER Macroeconomics Annual 2005, Volume 20, National Bureau of Economic Research, Inc, NBER Chapters, 313–382.
- MONACELLI, T. (2009): "New Keynesian models, durable goods, and collateral constraints," *Journal of Monetary Economics*, 56, 242–254.
- SCHMITT-GROHE, S. AND M. URIBE (2007): "Optimal inflation stabilization in a medium-scale macroeconomic model," in *Monetary Policy under Inflation Targeting*, ed. by F. S. Miskin, K. Schmidt-Hebbel, N. L. S. Editor), and K. S.-H. (Se, Central Bank of Chile, vol. 11 of *Central Banking*, *Analysis*, and *Economic Policies Book Series*, chap. 5, 125–186.

——— (2012): "What's news in business cycles," *Econometrica*, 80, 2733–2764.

- SHILLER, R. J. (2007): "Understanding recent trends in house prices and home ownership," NBER Working Papers 13553, National Bureau of Economic Research, Inc.
- SIMS, C. A. (2002): "Solving linear rational expectations models," Computational Economics, 20, 1–20.
- SMETS, F. R. AND R. WOUTERS (2003): "An estimated dynamic stochastic general equilibrium model of the euro area," *Journal of the European Economic Association*, 1, 1123–1175.
- (2007): "Shocks and frictions in US business cycles: a Bayesian DSGE approach," *American Economic Review*, 97, 586–606.
- STOCK, J. H. AND M. W. WATSON (1999): "Business cycle fluctuations in us macroeconomic time series," in *Handbook of Macroeconomics*, ed. by J. B. Taylor and M. Woodford, Elsevier, vol. 1 of *Handbook of Macroeconomics*, chap. 1, 3–64.

- TAYLOR, J. B. (1993): "Discretion versus policy rules in practice," Carnegie-Rochester Conference Series on Public Policy, 39, 195–214.
- YUN, T. (1996): "Nominal price rigidity, money supply endogeneity, and business cycles," *Journal of Monetary Economics*, 37, 345–370.

Parameter	Value	Parameter	Value
Households			
δ^h	0.01		
Lenders		Borrowers	
σ_P^x	1.2	σ_I^x	1.2
β_P	0.9925	β_I	0.9844
σ_P^x	1.2	σ^x_I	1.2
$ heta_P^x$	0.4	$ heta_I^x$	0.4
η_P	2.5	η_I	1.5
$ heta_P^n$	10.0	$ heta_I^n$	10.0
$\delta_0^{k^c}$	0.025		
$\delta_1^{k^c}$	0.0335		
${\delta_0^k}^h$	0.03		
$\delta_1^{k^h}$	0.0385		
Production			
α	0.25		
Consumption Sector		Housing Sector	
γ^c	0.25	γ^h	0.1
$ heta^c$	7.67	γ^l	0.35
$ heta^{n^c}$	7.67	$ heta^{n^h}$	7.67
Policy and Steady-Sta	ate		
ω	0.91	ϵ^h	0.75
π^c	1.005	ϵ^n	0.5
l	1.0	ϵ^{π}	1.005
u^{k^c}	1.0	ϵ^{R^m}	0.066
u^{k^h}	1.0	z^c	1.0
ϕ^m	20.0	z^h	1.0
ϵ^b	1.0	z^{i^k}	1.0
ϵ^{χ}	1.0		

 Table 1: CALIBRATED PARAMETERS

Variable	Model Definition			Data			Model
		Obs	Mean	Std	Min	Max	
Inflation Rate	$100\left(\left(\pi^c\right)^4 - 1\right)$	126	2.33	1.34	-0.53	6.41	2.00
Interest Rate							
Nominal Short-Term	$100(R^4 - 1)$	126	5.11	3.57	0.20	13.57	5.50
Nominal Long-Term	$100\left(\left(R^m\right)^4 - 1\right)$	126	8.06	2.91	3.99	14.47	8.41
Real Short-Term	$100\left(R^4 - (\pi^c)^4\right)$	126	2.77	2.99	-2.79	10.65	3.50
Flow as a Share of GDP							
Consumption	$100\left(\frac{c}{y}\right)$	126	77.35	1.44	75.02	80.46	74.25
Non-Housing Investment	$100\left(\frac{\left(q^{k^c}i^{k^c}+q^{k^h}i^{k^h}\right)}{y}\right)$	126	12.88	1.44	10.20	15.51	15.57
Housing Investment	$100\left(\frac{q^hy^h}{y}\right)$	126	9.77	1.11	8.04	12.56	7.88
Stock to GDP							
Capital	$\frac{\left(q^{k^c}k^c + q^{k^h}k^h\right)}{4y}$	30	1.44	0.10	1.30	1.66	1.55
Residential Structures	$rac{q^hy^h}{4y}$	30	1.24	0.12	1.05	1.46	1.97
Land	$rac{q^l l}{4y}$	30	0.91	0.20	0.66	1.32	0.81
Mortgage Debt	$rac{b_i}{4y}$	126	0.54	0.22	0.20	0.98	0.89
Hours Worked							
Consumption	$rac{n^c}{\left(n^c+n^h ight)}$	126	94.90	0.75	92.97	95.87	92.00
Housing	$rac{n^h}{\left(n^c+n^h ight)}$	126	5.10	0.75	4.13	7.03	8.00

Table 2: Steady-State Ratios

Note: Capital, Residential Structures, and Land values are represented at 2011 values due to a total reconstruction of the National Balance Sheet (Table 378-0049).

Parameter	Prior Distribution			Posterior Distribution				
	Distribution	Mean	Std	Mode	Std	5%	Median	95%
Households								
ξ^{w^c}	Beta	0.5	0.22	0.9797	0.0039	0.9740	0.9813	0.9870
ξ^{w^h}	Beta	0.5	0.22	0.9650	0.0068	0.9524	0.9653	0.9747
ι^{w^c}	Beta	0.5	0.22	0.3723	0.1018	0.1531	0.2579	0.4901
ι^{w^h}	Beta	0.5	0.22	0.8386	0.1545	0.4147	0.7094	0.9220
Lenders								
$\phi^{i^{k^c}}$	Gamma	5.0	2.0	3.1955	0.7956	2.3023	3.5164	4.8926
$\phi^{i^{k^h}}$	Gamma	5.0	2.0	7.7145	2.5711	4.5198	8.0855	12.9464
$\delta_2^{k^c}$	Beta	0.125	0.025	0.0619	0.0174	0.0407	0.0627	0.0966
$\delta_2^{k^h}$	Beta	0.125	0.025	0.1526	0.0245	0.1161	0.1529	0.1965
Production								
Consumptio	n Sector							
ξ^{p^c}	Beta	0.5	0.22	0.6226	0.0292	0.5573	0.6125	0.6528
ι^{p^c}	Beta	0.5	0.22	0.9598	0.0514	0.8203	0.9292	0.9833
Monetary P	olicy							
ρ_r	Beta	0.8	0.1	0.8604	0.0158	0.8189	0.8425	0.8701
$ ho_{\pi^c}$	Gamma	3.5	0.5	4.2291	0.3284	3.8153	4.3327	4.8643
$ ho_y$	Beta	0.30	0.025	0.3008	0.0199	0.2749	0.3075	0.3441

Table 3: PRIOR AND POSTERIOR DISTRIBUTIONS OF STRUCTURAL PARAMETERS

Note: As reported in the text, note that the parameters of the Taylor-type monetary policy rule are estimated on the sample covering 1992Q1 to 2014Q4 without any shock on inflation targeting. In the full sample estimation covering 1983Q3 to 2014Q4, the parameters of the Taylor-type monetary policy rule are calibrated and time-varying inflation targeting is introduced.

Parameter	Prior Dist	tribution	1		Poster	rior Distr	ibution	
	Distribution	Mean	Std	Mode	Std	5%	Median	95%
$ ho_{\epsilon\chi}$	Beta	0.8	0.1	0.8656	0.0212	0.8339	0.8695	0.9034
ρ_{ϵ^b}	Beta	0.8	0.1	0.9777	0.0257	0.9174	0.9740	0.9952
$ ho_{\epsilon^h}$	Beta	0.8	0.1	0.9526	0.0081	0.9406	0.9556	0.9677
$ ho_{\epsilon^n}$	Beta	0.8	0.1	0.9991	0.0005	0.9974	0.9985	0.9992
$ ho_{\epsilon^{\pi^c}}$	Beta	0.8	0.1	0.8718	0.0180	0.8361	0.8674	0.8945
$ ho_{\epsilon^{R^m}}$	Beta	0.8	0.1	0.9893	0.0108	0.9603	0.9820	0.9944
$ ho_{z^c}$	Beta	0.8	0.1	0.4836	0.0608	0.3593	0.4601	0.5624
ρ_{z^h}	Beta	0.8	0.1	0.9581	0.0269	0.8907	0.9521	0.9768
$\rho_{z^{ik}}$	Beta	0.8	0.1	0.6819	0.0542	0.5650	0.6516	0.7447
σ_{χ}	Inv. Gamma	0.1	0.2	0.0272	0.0018	0.0247	0.0274	0.0307
σ_{ϵ^b}	Inv. Gamma	0.1	0.2	0.0171	0.0060	0.0114	0.0164	0.0287
$\sigma_{\epsilon^h,0}$	Inv. Gamma	0.1	0.2	0.0072	0.0009	0.0060	0.0073	0.0087
$\sigma_{\epsilon^h,4}$	Inv. Gamma	0.1	0.2	0.0044	0.0005	0.0037	0.0044	0.0052
$\sigma_{\epsilon^h,8}$	Inv. Gamma	0.1	0.2	0.0045	0.0005	0.0038	0.0045	0.0054
σ_{ϵ^n}	Inv. Gamma	0.1	0.2	0.0277	0.0061	0.0272	0.0360	0.0482
$\sigma_{\epsilon^{\pi^c}}$	Inv. Gamma	0.1	0.2	0.0034	0.0003	0.0031	0.0035	0.0040
σ_{ϵ^R}	Inv. Gamma	0.1	0.2	0.0035	0.0003	0.0032	0.0036	0.0041
$\sigma_{\epsilon^{R^m}}$	Inv. Gamma	0.1	0.2	0.1374	0.0170	0.0953	0.1229	0.1511
σ_{z^c}	Inv. Gamma	0.1	0.2	0.0090	0.0011	0.0072	0.0089	0.0109
σ_{z^h}	Inv. Gamma	0.1	0.2	0.0162	0.0010	0.0146	0.0161	0.0180
$\sigma_{z^{ik}}$	Inv. Gamma	0.1	0.2	0.0500	0.0136	0.0368	0.0567	0.0811

Table 4: PRIOR AND POSTERIOR DISTRIBUTIONS OF EXOGENOUS PROCESSES

Parameter	Prior Dist	tributior	1		Poster	rior Distr	ibution	
	Distribution	Mean	Std	Mode	Std	5%	Median	95%
σ_{n^c}	Inv. Gamma	0.1	0.2	0.0164	0.0015	0.0148	0.0170	0.0196
σ_{w^c}	Inv. Gamma	0.1	0.2	0.0123	0.0008	0.0112	0.0125	0.0140
$\sigma_{u^{k^c}}$	Inv. Gamma	0.1	0.2	0.0379	0.0025	0.0344	0.0381	0.0426
σ_{n^h}	Inv. Gamma	0.1	0.2	0.0076	0.0005	0.0069	0.0077	0.0085
σ_{w^h}	Inv. Gamma	0.1	0.2	0.0147	0.0010	0.0131	0.0147	0.0165
$\sigma_{u^{k^h}}$	Inv. Gamma	0.1	0.2	0.0650	0.0041	0.0589	0.0650	0.0725

Table 5: PRIOR AND POSTERIOR DISTRIBUTIONS OF MEASUREMENT ERRORS

Correlations	Prior Dis	stribution	l	Posterior Distribution
	Distribution	Mean	Std	Mode
$corr\left(c_{t},y_{t}^{h} ight)$	Normal	0.6590	0.1	0.5581
$corr\left(c_{t}, y_{t-1}^{h}\right)$	Normal	0.6811	0.1	0.5525
$corr\left(c_{t}, y_{t-2}^{h}\right)$	Normal	0.6589	0.1	0.5603
$corr\left(c_{t}, y_{t-3}^{h}\right)$	Normal	0.6204	0.1	0.5666
$corr\left(c_{t-1}, y_{t}^{h}\right)$	Normal	0.5493	0.1	0.5344
$corr\left(c_{t-2}, y_{t}^{h}\right)$	Normal	0.4279	0.1	0.5304
$corr\left(c_{t-3}, y_{t}^{h}\right)$	Normal	0.3185	0.1	0.5305
$corr\left(c_{t},q_{t}^{h} ight)$	Normal	0.5074	0.1	0.4330
$corr\left(c_{t},q_{t-1}^{h}\right)$	Normal	0.5256	0.1	0.4378
$corr\left(c_{t},q_{t-2}^{h} ight)$	Normal	0.4766	0.1	0.4491
$corr\left(c_{t},q_{t-3}^{h} ight)$	Normal	0.4043	0.1	0.4579
$corr\left(c_{t-1}, q_{t}^{h}\right)$	Normal	0.4634	0.1	0.3949
$corr\left(c_{t-2}, q_{t}^{h}\right)$	Normal	0.4335	0.1	0.3874
$corr\left(c_{t-3}, q_{t}^{h}\right)$	Normal	0.4079	0.1	0.3862
$corr\left(c_{t},i_{t}^{k} ight)$	Normal	0.6713	0.1	0.5359
$corr\left(c_{t}, i_{t-1}^{k}\right)$	Normal	0.6293	0.1	0.5367
$corr\left(c_{t}, i_{t-2}^{k}\right)$	Normal	0.5508	0.1	0.5445
$corr\left(c_{t}, i_{t-3}^{k}\right)$	Normal	0.4747	0.1	0.5552
$corr\left(c_{t-1}, i_{t}^{k}\right)$	Normal	0.6937	0.1	0.5519
$corr\left(c_{t-2},i_{t}^{k} ight)$	Normal	0.6498	0.1	0.5612
$corr\left(c_{t-3},i_{t}^{k} ight)$	Normal	0.5695	0.1	0.5692
$corr\left(c_{t},M_{t} ight)$	Normal	0.3335	0.1	0.7881
$corr\left(c_{t},M_{t-1}\right)$	Normal	0.2441	0.1	0.5034
$corr\left(c_{t},M_{t-2}\right)$	Normal	0.1474	0.1	0.4505
$corr\left(c_{t},M_{t-3} ight)$	Normal	0.0514	0.1	0.4359
$corr\left(c_{t-1},M_{t}\right)$	Normal	0.4231	0.1	0.4480
$corr\left(c_{t-2},M_{t}\right)$	Normal	0.4871	0.1	0.3857
$corr\left(c_{t-3},M_{t}\right)$	Normal	0.5306	0.1	0.3706

 Table 6: MODEL PRIORS

Tal	ole 7: Selected Auto	OCORRELAT	TON AND CRO	DSS-CORRI	ELATION	
Moments		Autoo	correlation and Cro	ss-Correlation	of Order 0	
	Variables	Consumption	Non-Residential Investment	Residential Investment	Mortgage Debt	House Price
Sample	Consumption	1.00	0.67	0.66	0.33	0.51
A symptotic		1.00	0.54	0.56	0.79	0.43
Sample	Non-Residential Investment	0.67	1.00	0.40	0.36	0.37
A symptotic		0.54	1.00	0.64	0.39	0.58
Sample	Residential Investment	0.66	0.40	1.00	0.13	0.56
A symptotic		0.56	0.64	1.00	0.54	0.72
Sample	Mortgage Debt	0.33	0.36	0.13	1.00	0.63
A symptotic		0.79	0.39	0.54	1.00	0.57
Sample	House Price	0.51	0.37	0.56	0.63	1.00
Asymptotic		0.43	0.58	0.72	0.57	1.00
Moments		Autoo	correlation and Cro	ss-Correlation	of Order 1	
	Variables	Consumption	Non-Residential Investment	Residential Investment	Mortgage Debt	House Price
Sample	Consumption	0.91	0.69	0.55	0.42	0.46
$\operatorname{Asymptotic}$		0.78	0.55	0.53	0.45	0.39
Sample	Non-Residential Investment	0.63	0.93	0.28	0.40	0.31
Asymptotic		0.54	0.99	0.65	0.41	0.60
Sample	Residential Investment	0.68	0.48	0.90	0.24	0.54
A symptotic		0.55	0.63	0.98	0.53	0.72
Sample	Mortgage Debt	0.24	0.30	0.02	0.99	0.57
A symptotic		0.50	0.40	0.50	0.54	0.52
Sample	House Price	0.53	0.44	0.53	0.69	0.84
Asymptotic		0.44	0.57	0.71	0.58	0.98
Moments		Autoo	correlation and Cro	ss-Correlation	of Order 4	
	Variables	Consumption	Non-Residential	Residential	Mortgage	House
			Investment	Investment	Debt	Price
\mathbf{Sample}	Consumption	0.60	0.48	0.22	0.56	0.39
A symptotic		0.71	0.58	0.53	0.36	0.39
\mathbf{Sample}	Non-Residential Investment	0.39	0.49	0.02	0.45	0.25
Asymptotic		0.57	0.88	0.66	0.48	0.62
Sample	Residential Investment	0.55	0.50	0.38	0.43	0.51
A symptotic		0.57	0.61	0.92	0.54	0.70
Sample	Mortgage Debt	-0.04	0.09	-0.27	0.87	0.40
A symptotic		0.43	0.40	0.48	0.40	0.48
Sample	House Price	0.31	0.39	0.12	0.77	0.59
Asymptotic		0.46	0.55	0.66	0.59	0.92

ζ t < ζ 1 Table

Table 0. 0100													
Observables	ε ^{εχ}	$\varepsilon^{\epsilon^{b}}$	$\varepsilon^{\epsilon^{h,0}}$	$\varepsilon^{\epsilon^{h,4}}$	$\varepsilon^{\epsilon^{h,8}}$	ε _e z	ε ^{eπc}	ε^{ϵ_R}	$\varepsilon^{\epsilon_{R^m}}$	ezc	Ezh	$\varepsilon^{z^{ik}}$	m.e.
Consumption	18.02	0.24	6.06	2.32	1.93	54.00	9.80	3.29	2.64	0.19	0.96	0.55	0.00
Capital Price	24.90	0.73	9.48	7.76	7.64	1.01	3.19	3.56	3.20	2.41	1.61	34.51	0.00
Non-Residential Investment	4.99	5.34	5.19	2.73	4.63	39.58	12.06	0.99	0.35	0.27	0.53	23.36	0.00
House Price	0.95	1.23	20.32	8.07	8.40	24.15	2.81	0.62	17.22	0.16	14.63	1.46	0.00
Residential Investment	0.46	0.51	11.16	4.81	5.56	41.13	2.94	0.49	11.34	0.09	21.20	0.30	0.00
Mortgage Debt	35.32	0.49	14.72	2.60	2.00	14.83	8.53	4.91	14.31	0.13	1.66	0.50	0.00
Core CPI Inflation	2.02	0.30	0.67	0.11	0.08	90.84	2.52	0.44	0.24	2.09	0.11	0.58	0.00
Treasury Bill Rate, 3 month	12.15	0.68	4.08	1.50	1.09	74.17	2.84	0.66	1.29	0.54	0.65	0.35	0.00
Residential Mortgage Rate, 5 year	0.55	0.73	0.39	0.25	0.31	89.10	0.32	0.14	7.99	0.05	0.09	0.09	0.00
Hours Worked (Consumption)	35.71	0.08	10.80	1.57	1.03	24.92	11.90	6.59	3.77	0.72	1.46	0.88	0.57
Wage (Consumption)	3.99	12.58	10.63	3.38	2.64	4.71	9.42	0.61	1.48	4.45	0.68	19.32	26.10
Capacity Utilization Rate (Consumption)	37.13	1.86	12.48	2.47	1.66	1.14	9.52	5.40	3.89	0.72	1.60	2.94	19.19
Hours Worked (Housing)	0.54	0.42	13.19	5.61	6.39	42.38	3.73	0.64	12.25	0.12	5.24	0.29	9.19
Wage (Housing)	1.89	6.85	11.24	3.88	3.30	18.55	7.26	0.32	6.36	3.92	1.50	15.92	19.02
Capacity Utilization Rate (Housing)	0.14	0.41	4.70	1.83	1.78	1.19	1.23	0.18	2.12	0.02	1.85	0.35	84.19

DECOMPOSITION
VARIANCE
Error
FORECAST
JNCONDITIONAL
ole 8: 1
Tat

Figure 1: DATA FROM 1983Q3 TO 2014Q4





Figure 2: ESTIMATED (SMOOTHED) SHOCKS FROM 1983Q3 TO 2014Q4



Figure 3: Selected Historical Variance Decomposition



Figure 4: SHOCK RESPONSE FUNCTION – HOUSING DEMAND SHOCKS



Figure 5: SHOCK RESPONSE FUNCTION – LABOUR SUPPLY SHOCK



Figure 6: SHOCK RESPONSE FUNCTION – INFLATION TARGETING AND MONETARY POLICY SHOCKS



Figure 7: SHOCK RESPONSE FUNCTION – MORTGAGE PREMIUM SHOCK







Figure 9: Anticipated and Unanticipated Housing Demand Shock



Figure 10: COUNTERCYCLICAL LTV POLICY BASED ON HOUSE PRICE



9

S

0

7

Quarters

S

2

-0.2

Quarters

S

7

Quarters

Figure 11: COUNTERCYCLICAL LTV POLICY BASED ON DEBT-TO-GDP RATIO



Figure 12: MONETARY POLICY RESPONSIVE TO HOUSE PRICE INFLATION

Appendices

A Model

A.1 Households

Patient The Lagrangian for the patient optimization problem takes the following form:

$$\begin{split} L &= E_0 \sum_{t=0}^{\infty} \beta_P^t \epsilon_t^b \times \\ & \left\{ \begin{array}{l} U\left(c_{P,t}, h_{P,t}, n_{P,t}^c, n_{P,t}^b\right) + \lambda_{P,t}^c \times \\ \left\{ \begin{array}{l} n_{P,t}^{c,d} \int_0^1 w_{P,e,t}^c \left(\frac{w_{P,e,t}^c}{w_{P,t}^c}\right)^{-\theta^{n^c}} de + n_{P,t}^{h,d} \int_0^1 w_{P,e,t}^h \left(\frac{w_{P,e,t}^h}{w_{P,t}^b}\right)^{-\theta^{n^h}} de \\ + r_t^{k^c} u_t^{k^c} k_{t-1}^c + r_t^{k^h} u_t^{k^h} k_{t-1}^h + \left(q_t^l + r_t^l\right) l_{t-1} + \frac{R_{t-1}b_{P,t-1}}{\pi_t^c} + f_t^c + f_t^h + f_t^{fi} \\ + \frac{1}{\phi^m} \sum_{s=1}^{\infty} \frac{1}{\Pi_{v=-s}^{0-s} \pi_{t+v}^c} + \sum_{s=1}^{\phi^m} \left(R_{t-s}^d - 1\right) \left(\frac{\phi^{m-s+1}}{\phi^m}\right) \frac{d_{t-s}}{\Pi_{v=-s}^{0-s} \pi_{t+v}^c} \\ - c_{P,t} - q_t^{k^c} i_t^{k^c} - q_t^{k^h} i_t^{k^h} - q_t^l l_t - b_{P,t} - d_t \\ - q_t^h \left[h_{P,t} - \left(1 - \delta_t^h\right) h_{P,t-1}\right] \\ - q_t^{k^c} \left[k_t^c - \left(1 - \delta_t^{k^h}\right) k_{t-1}^h - z_t^{i^k} i_t^{k^h} \left[1 - \frac{\phi^{k^c}}{2} \left(\frac{i_t^{k^h}}{i_{t-1}^{k-1}} - 1\right)^2\right]\right] \\ - q_t^{k^h} \left[k_t^h - \left(1 - \delta_t^{k^h}\right) k_{t-1}^h - z_t^{i^k} i_t^{k^h} \left[1 - \frac{\phi^{k^h}}{2} \left(\frac{i_t^{k^h}}{i_{t-1}^{k-1}} - 1\right)^2\right]\right] \\ + \frac{w_{P,t}^{r_{t,t}}}{\lambda_{P,t}^{r_{t,t}}} \left[n_{P,t}^c - n_{P,t}^{c,d} \int_0^1 \left(\frac{w_{P,e,t}^c}{w_{P,t}^r}\right)^{-\theta^{n^c}} de\right] + \frac{w_{P,t}^h}{\lambda_{P,t}^{h,t}} \left[n_{P,t}^h - n_{P,t}^{h,d} \int_0^1 \left(\frac{w_{P,e,t}^h}{w_{P,t}^h}\right)^{-\theta^{n^h}} de\right] \end{array} \right\} \end{split}$$

The first-order necessary conditions for $c_{P,t}$, $h_{P,t}$, $n_{P,t}^c$, $n_{P,t}^h$, k_t^c , k_t^h , l_t , $u_t^{k^c}$, $u_t^{k^h}$, $i_t^{k^c}$, $i_t^{k^h}$, $b_{P,t}$ and d_t are, respectively,

$$\lambda_{P,t}^{c} = \frac{\left(1 - \epsilon_{t}^{h}\right)^{\frac{1}{\theta_{P}^{c}}} c_{P,t}^{\frac{-1}{\theta_{P}^{c}}}}{x_{P,t}^{\frac{\sigma_{P}^{x} \theta_{P}^{x} - 1}{\theta_{P}^{c}}},$$
(36)
$$\epsilon_{t}^{b}\lambda_{P,t}^{c}q_{t}^{h} - \beta_{P}\left(1 - \delta^{h}\right)E_{t}\left[\epsilon_{t+1}^{b}\lambda_{P,t+1}^{c}q_{t+1}^{h}\right] = \epsilon_{t}^{b}\frac{\left(\epsilon_{t}^{h}\right)^{\frac{1}{\theta_{P}}}h_{P,t}^{\frac{1}{\theta_{P}}}}{x_{P,t}^{\frac{\sigma_{P}^{\sigma}\theta_{P}^{-1}}{\theta_{P}}}},$$
(37)

$$\epsilon_{t}^{n} \left(\left(n_{P,t}^{c} \right)^{\frac{\theta_{P}^{n+1}}{\theta_{P}^{n}}} + \left(n_{P,t}^{h} \right)^{\frac{\theta_{P}^{n+1}}{\theta_{P}^{n}}} \right)^{\frac{\theta_{P}^{n}\eta_{P}-1}{\theta_{P}^{n+1}}} \left(n_{P,t}^{c} \right)^{\frac{1}{\theta_{P}^{n}}} = \frac{\lambda_{P,t}^{c} w_{P,t}^{c}}{\lambda_{P,t}^{n^{c}}}, \tag{38}$$

$$\epsilon_t^n \left(\left(n_{P,t}^c \right)^{\frac{\theta_P^n + 1}{\theta_P^n}} + \left(n_{P,t}^h \right)^{\frac{\theta_P^n + 1}{\theta_P^n}} \right)^{\frac{\theta_P^n + 1}{\theta_P^n + 1}} \left(n_{P,t}^h \right)^{\frac{1}{\theta_P^n}} = \frac{\lambda_{P,t}^c w_{P,t}^h}{\lambda_{P,t}^{n^h}}, \tag{39}$$

$$\epsilon_t^b \lambda_{P,t}^c q_t^{k^c} = \beta_P E_t \left[\epsilon_{t+1}^b \lambda_{P,t+1}^c \left(u_{t+1}^{k^c} r_{t+1}^{k^c} + q_{t+1}^{k^c} \left(1 - \delta_{t+1}^{k^c} \right) \right) \right], \tag{40}$$

$$\epsilon_t^b \lambda_{P,t}^c q_t^{k^h} = \beta_P E_t \left[\epsilon_{t+1}^b \lambda_{P,t+1}^c \left(u_{t+1}^{k^h} r_{t+1}^{k^h} + q_{t+1}^{k^h} \left(1 - \delta_{t+1}^{k^h} \right) \right) \right], \tag{41}$$

$$\epsilon_t^b \lambda_{P,t}^c q_t^l = \beta_P E_t \left[\epsilon_{t+1}^b \lambda_{P,t+1}^c \left(q_{t+1}^l + r_{t+1}^l \right) \right], \tag{42}$$

$$r_t^{k^c} = q_t^{k^c} \left[\delta_1^{k^c} + \delta_2^{k^c} \left(u_t^{k^c} - 1 \right) \right], \tag{43}$$

$$r_t^{k^h} = q_t^{k^h} \left[\delta_1^{k^h} + \delta_2^{k^h} \left(u_t^{k^h} - 1 \right) \right], \tag{44}$$

$$\epsilon_{t}^{b}\lambda_{P,t}^{c}\left(1-q_{t}^{k^{c}}z_{t}^{i^{k}}\left(1-\frac{\phi^{k^{c}}}{2}\left(\frac{i_{t}^{k^{c}}}{i_{t-1}^{k^{c}}}-1\right)^{2}-\phi^{k^{c}}\left(\frac{i_{t}^{k^{c}}}{i_{t-1}^{k^{c}}}-1\right)\frac{i_{t}^{k^{c}}}{i_{t-1}^{k^{c}}}\right)\right)=\beta_{P}E_{t}\left[\epsilon_{t+1}^{b}\lambda_{P,t+1}^{c}q_{t+1}^{k^{c}}z_{t+1}^{i^{k}}\phi^{k^{c}}\left(\frac{i_{t+1}^{k^{c}}}{i_{t}^{k^{c}}}-1\right)\left(\frac{i_{t+1}^{k^{c}}}{i_{t}^{k^{c}}}\right)^{2}\right],\quad(45)$$

$$\epsilon_{t}^{b}\lambda_{P,t}^{c}\left(1-q_{t}^{k^{h}}z_{t}^{i^{k}}\left(1-\frac{\phi^{k^{h}}}{2}\left(\frac{i_{t}^{k^{h}}}{i_{t-1}^{k^{h}}}-1\right)^{2}-\phi^{k^{h}}\left(\frac{i_{t}^{k^{h}}}{i_{t-1}^{k^{h}}}-1\right)\frac{i_{t}^{k^{h}}}{i_{t-1}^{k^{h}}}\right)\right)=$$

$$\beta_{P}E_{t}\left[\epsilon_{t+1}^{b}\lambda_{P,t+1}^{c}q_{t+1}^{k^{h}}z_{t+1}^{i^{k}}\phi^{k^{h}}\left(\frac{i_{t+1}^{k^{h}}}{i_{t}^{k^{h}}}-1\right)\left(\frac{i_{t+1}^{k^{h}}}{i_{t}^{k^{h}}}\right)^{2}\right],\quad(46)$$

$$\epsilon_t^b \lambda_{P,t}^c = \beta_P R_t E_t \left[\epsilon_{t+1}^b \frac{\lambda_{P,t+1}^c}{\pi_{t+1}^c} \right], \qquad (47)$$

and

$$\epsilon_t^b \lambda_{P,t}^c = \sum_{s=1}^{\phi^m} \beta_P^s E_t \left[\epsilon_{t+s}^b \frac{\lambda_{P,t+s}^c}{\prod_{v=1}^s \pi_{t+v}^c} \left(\frac{1}{\phi^m} + \left(R_t^d - 1 \right) \left(\frac{\phi^m - s + 1}{\phi^m} \right) \right) \right],\tag{48}$$

where $\lambda_{P,t}^c$ is the Lagrange multiplier on budget constraint (7), $\frac{\lambda_{P,t}^c w_{P,t}^j}{\lambda_{P,t}^{n_j}}$ and $\lambda_{P,t}^c q_t^{k^j}$ are the Lagrange multipliers on labour supply constraints (6) and the law of motion of capital (8), respectively. Equation (36) describes the marginal utility of current consumption of non-durable goods. Equation (37) requires that households equate the marginal utility of current consumption goods to the marginal utility increase of housing stock services, the latter being composed of two parts: (i) the direct utility gain of an additional unit of housing, and (ii) the expected utility stemming from the consumption of the resale value of housing purchased in previous periods. Equations (38) and (39) link real wages in both sectors to households' marginal rate of substitution between consumption goods and leisure. In equilibrium, real wages in the consumption and housing sectors are equal. Equations (40) and (41) require that households equate their marginal utility of current consumption goods to the marginal utility increase of an additional unit of capital, which includes two parts: (i) the rental rate of capital, and (ii) the expected utility stemming from the consumption of the resale value of undepreciated capital purchased in previous periods. Equations (43) and (44) link the variable capacity utilization rate to the rental rate of capital. Equations (45) and (46) require that households equate the investment cost, in terms of the foregone marginal utility of consumption goods, to the expected value of the rebate in adjustment cost in the following period. Equation (47) is the typical Euler condition that equates the cost of sacrificing one unit of consumption goods to the benefit of investing in the bond market.³⁸ Finally, equation (48) equates the cost of sacrificing one unit of consumption goods to the benefit of making deposits generating a flow of revenues for ϕ^m periods.

 $^{^{38}{\}rm Since}$ lenders own all firms and financial intermediairies, it also determines the pricing kernel of the economy.

Impatient The Lagrangian for the impatient optimization problem takes the following form:

$$\begin{split} L &= E_0 \sum_{t=0}^{\infty} \beta_I^t \epsilon_t^b \times \\ & \left\{ \begin{array}{l} U\left(c_{I,t}, h_{I,t}, n_{I,t}^c, n_{I,t}^h\right) + \lambda_{I,t}^c \times \\ \left\{ \begin{array}{l} n_{I,t}^{c,d} \int_0^1 w_{I,e,t}^c \left(\frac{w_{I,e,t}^c}{w_{I,t}^c}\right)^{-\theta^{n^c}} de + n_{I,t}^{h,d} \int_0^1 w_{I,t}^h \left(\frac{w_{I,e,t}^h}{w_{I,t}^h}\right)^{-\theta^{n^h}} de \\ + \frac{R_{t-1}^{m}b_{I,t-1}}{\pi_t^c} + m_t - c_{I,t} - b_{I,t} - q_t^h \left[h_{I,t} - (1 - \delta^h) h_{I,t-1}\right] \\ - \frac{1}{\phi^m} \sum_{s=1}^{\phi^m} \frac{m_{t-s}}{\prod_{v=-s}^0 \pi_{t+v}^c} - \sum_{s=1}^{\phi^m} \left(R_{t-s}^m - 1\right) \left(\frac{\phi^m - s + 1}{\phi^m}\right) \frac{m_{t-s}}{\prod_{v=-s}^0 \pi_{t+v}^c} \\ + \frac{w_{I,t}^c}{\lambda_{I,t}^{n^c}} \left[n_{I,t}^c - n_{I,t}^{c,d} \int_0^1 \left(\frac{w_{I,e,t}^c}{w_{I,t}^c}\right)^{-\theta^{n^c}} de \right] + \frac{w_{I,t}^h}{\lambda_{I,t}^{n^h}} \left[n_{I,t}^h - n_{I,t}^{h,d} \int_0^1 \left(\frac{w_{I,e,t}^h}{w_{I,t}^h}\right)^{-\theta^{n^h}} de \right] \\ + \lambda_{I,t}^b \left[M_t + \omega h_{I,t} q_t^h\right] \end{split} \right\} \end{split}$$

The first-order necessary conditions for $c_{I,t}$, $h_{I,t}$, $n_{I,t}^c$, $n_{I,t}^h$, $b_{I,t}$ and m_t are, respectively,

$$\lambda_{I,t}^{c} = \frac{\left(1 - \epsilon_{t}^{h}\right)^{\frac{1}{\theta_{T}^{c}}} c_{I,t}^{\frac{-1}{\theta_{T}^{c}}}}{\frac{\sigma_{I}^{x} \theta_{T}^{x} - 1}{\theta_{T}^{x}}},\tag{49}$$

$$\epsilon_t^b \lambda_{I,t}^c q_t^h - \beta_I \left(1 - \delta^h \right) E_t \left[\epsilon_{t+1}^b \lambda_{I,t+1}^c q_{t+1}^h \right] = \epsilon_t^b \frac{\left(\epsilon_t^h \right)^{\frac{1}{\theta_I^x}} h_{I,t}^{\frac{-1}{\theta_I^x}}}{x_{I,t}^{\frac{\sigma_I^x \theta_I^x - 1}{\theta_I^x}}} + \epsilon_t^b \lambda_{I,t}^c \lambda_t^b \omega q_t^h, \tag{50}$$

$$\epsilon_t^n \left(\left(n_{I,t}^c \right)^{\frac{\theta_I^n + 1}{\theta_I^n}} + \left(n_{I,t}^h \right)^{\frac{\theta_I^n + 1}{\theta_I^n}} \right)^{\frac{\theta_I^n \eta_I - 1}{\theta_I^n + 1}} \left(n_{I,t}^c \right)^{\frac{1}{\theta_I^n}} = \frac{\lambda_{I,t}^c w_{I,t}^c}{\lambda_{I,t}^{n^c}}, \tag{51}$$

$$\epsilon_{t}^{n} \left(\left(n_{I,t}^{c} \right)^{\frac{\theta_{I}^{n}+1}{\theta_{I}^{n}}} + \left(n_{I,t}^{h} \right)^{\frac{\theta_{I}^{n}+1}{\theta_{I}^{n}}} \right)^{\frac{\theta_{I}^{n}\eta_{I}-1}{\theta_{I}^{n}+1}} \left(n_{I,t}^{h} \right)^{\frac{1}{\theta_{I}^{n}}} = \frac{\lambda_{I,t}^{c} w_{I,t}^{h}}{\lambda_{I,t}^{n^{h}}}, \tag{52}$$

$$\epsilon_t^b \lambda_{I,t}^c = \beta_I R_t E_t \left[\epsilon_t^b \frac{\lambda_{I,t+1}^c}{\pi_{t+1}^c} \right], \tag{53}$$

and

$$\epsilon_t^b \lambda_{I,t}^c \left(1 - \lambda_t^b \right) = \sum_{s=1}^{\phi^m} \beta_I^s E_t \left[\epsilon_{t+s}^b \frac{\lambda_{I,t+s}^c}{\prod_{v=1}^s \pi_{t+v}^c} \left(\frac{1}{\phi^m} + (R_t^m - 1) \left(\frac{\phi^m - s + 1}{\phi^m} \right) + \lambda_{t+s}^b \left(\frac{\phi^m - s}{\phi^m} \right) \right) \right]$$
(54)

where $\lambda_{I,t}^c$ is the Lagrange multiplier on budget constraint (10), $\frac{\lambda_{I,t}^c w_{I,t}^j}{\lambda_{I,t}^{nj}}$ and $\lambda_{I,t}^c \lambda_{I,t}^b$ are the Lagrange multipliers on labour supply constraint (6) and the borrowing constraint (11), respectively. Equations (49), (51), (52) and (53) have the same interpretation as for the lenders. Finally, equations (50) and (54) depend on the same two components as the lenders' equations, but also on the marginal utility of relaxing the borrowing constraint.

Wages The Lagrangian for wages optimization problem (for $i \in \{P, I\}$ and $j \in \{c, h\}$) takes the following form:

$$L = E_t \sum_{s=0}^{\infty} \left(\beta_i \xi^{w^j}\right)^s \epsilon_{t+s}^b \lambda_{i,t+s}^c n_{i,t+s}^{j,d} \left(\frac{\tilde{w}_{i,t}^j}{w_{i,t+s}^j}\right)^{-\theta^{n^j}} \prod_{k=1}^s \left(\frac{(\pi^c)^{\iota^{w^j}} (\pi_{t+k-1}^c)^{1-\iota^{w^j}}}{\pi_{t+k}^c}\right)^{-\theta^{n^j}} \times \left[\tilde{w}_{i,t}^j \prod_{k=1}^s \left(\frac{(\pi^c)^{\iota^{w^j}} (\pi_{t+k-1}^c)^{1-\iota^{w^j}}}{\pi_{t+k}^c}\right) - \frac{w_{i,t+s}^j}{\lambda_{i,t+s}^{n^j}}\right]$$

The households' first-order necessary condition with respect to the optimally set wage rate in the current period in the production sector j, $\tilde{w}_{i,t}^{j}$, is

$$E_{t} \sum_{s=0}^{\infty} \left(\beta_{i} \xi^{w^{j}}\right)^{s} \epsilon_{t+s}^{b} \lambda_{i,t+s}^{c} n_{i,t+s}^{j,d} \left(\frac{\tilde{w}_{i,t}^{j}}{w_{i,t+s}^{j}}\right)^{-\theta^{n^{j}}} \prod_{k=1}^{s} \left(\frac{\left(\pi^{c}\right)^{\iota^{w^{j}}} \left(\pi_{t+k-1}^{c}\right)^{1-\iota^{w^{j}}}}{\pi_{t+k}^{c}}\right)^{-\theta^{n^{j}}} \times \left[\frac{\theta^{n^{j}} - 1}{\theta^{n^{j}}} \tilde{w}_{i,t}^{j}}{\theta^{n^{j}}} \prod_{k=1}^{s} \frac{\left(\pi^{c}\right)^{\iota^{w^{j}}} \left(\pi_{t+k-1}^{c}\right)^{1-\iota^{w^{j}}}}{\pi_{t+k}^{c}} - \frac{w_{i,t}^{j}}{\lambda_{i,t}^{n^{j}}}\right] = 0.$$

Using (38), (39), (51) and (52) to eliminate $\lambda_{i,t}^{n^j}$ yields

$$E_{t} \sum_{s=0}^{\infty} \left(\beta_{i} \xi^{w^{j}}\right)^{s} \epsilon_{t+s}^{b} \lambda_{i,t+s}^{c} n_{i,t+s}^{j,d} \left(\frac{\tilde{w}_{i,t}^{j}}{w_{i,t+s}^{j}}\right)^{-\theta^{n^{j}}} \prod_{k=1}^{s} \left(\frac{\left(\pi^{c}\right)^{\iota^{w^{j}}} \left(\pi_{t+k-1}^{c}\right)^{1-\iota^{w^{j}}}}{\pi_{t+k}^{c}}\right)^{-\theta^{n^{j}}} \times \left[\frac{\theta^{n^{j}}-1}{\theta^{n^{j}}} \tilde{w}_{i,t}^{j}}{\prod_{k=1}^{s} \frac{\left(\pi^{c}\right)^{\iota^{w^{j}}} \left(\pi_{t+k-1}^{c}\right)^{1-\iota^{w^{j}}}}{\pi_{t+k}^{c}}} - \frac{1}{\lambda_{i,t+s}^{c}} \epsilon_{t+s}^{n} \left(\left(n_{i,t+s}^{c}\right)^{\frac{\theta^{n+1}}{\theta^{n}}} + \left(n_{i,t+s}^{h}\right)^{\frac{\theta^{n+1}}{\theta^{n}}}\right)^{\frac{\theta^{n}_{i}+1}{\theta^{n}_{i}}} \left(n_{i,t+s}^{j}\right)^{\frac{1}{\theta^{n}}}\right] = 0$$

This equation states that, in labour markets in which the wage rate is reoptimized in period t, the real wage is set to equate the expected future average marginal revenue to the average marginal cost of supplying labour. In this equation, $\frac{\theta^{n^j}}{\theta^{n^j}-1}$ represents the markup of wages over the marginal cost of labour that would prevail in the absence of wage stickiness and trend inflation. To write the wage-setting equation in recursive form,³⁹ we need to define intermediate variables $f_{i,t}^{1,j}$ and $f_{i,t}^{2,j}$. This yields

$$\begin{split} f_{i,t}^{1,j} &= \epsilon_t^b \frac{\theta^{n^j} - 1}{\theta^{n^j}} \left(\tilde{w}_{i,t}^j \right)^{1 - \theta^{n^j}} \left(\frac{1}{w_{i,t}^j} \right)^{-\theta^{n^j}} \lambda_{i,t}^c n_{i,t}^{j,d} + \\ & \beta_i \xi^{w^j} E_t \left[\left(\frac{\left(\pi^c \right)^{\iota^{w^j}} \left(\pi_t^c \right)^{1 - \iota^{w^j}}}{\pi_{t+1}^c} \right)^{1 - \theta^{n^j}} \left(\frac{\tilde{w}_{i,t}^j}{\tilde{w}_{i,t+1}^j} \right)^{1 - \theta^{n^j}} f_{i,t+1}^{1,j} \right], \\ f_{i,t}^{2,j} &= \epsilon_t^b \epsilon_t^n \left(\left(n_{i,t}^c \right)^{\frac{\theta^{n+1}}{\theta^n}} + \left(n_{i,t}^h \right)^{\frac{\theta^{n+1}}{\theta^n}} \right)^{\frac{\theta^n + 1}{\theta^n + 1}} \left(n_{i,t}^j \right)^{\frac{1}{\theta^n}} \left(\frac{\tilde{w}_{i,t}^j}{w_{i,t}^j} \right)^{-\theta^{n^j}} n_{i,t}^{j,d} + \\ & \beta_i \xi^{w^j} E_t \left[\left(\frac{\left(\pi^c \right)^{\iota^{w^j}} \left(\pi_t^c \right)^{1 - \iota^{w^j}}}{\pi_{t+1}^c} \right)^{-\theta^{n^j}} \left(\frac{\tilde{w}_{i,t}^j}{\tilde{w}_{i,t+1}^j} \right)^{-\theta^{n^j}} f_{i,t+1}^{2,j} \right], \\ f_{i,t}^{1,j} &= f_{i,t}^{2,j}. \end{split}$$

³⁹Which is necessary for the representation of the model in state-space form.

A.2 Firms

A.2.1 Consumption Sector

Final-Goods Producers For any given level of final consumption goods produced, final-goods producers must solve the expenditure-minimizing problem

$$\min_{\{y_{m,t}^c\}} \int_{0}^{1} P_{m,t}^c y_{m,t}^c dm$$

subject to aggregation constraint (13).

Intermediate-Goods Producers The nominal profits (i.e. dividends) of the firm are denoted by

$$F_{m,t}^c = P_{m,t}^c y_{m,t}^c - R_t^{k^c} k_{m,t}^c - W_{P,t}^c n_{P,m,t}^{c,d} - W_{I,t}^c n_{I,m,t}^{c,d}.$$

The firm's objective is a static problem of profit maximization

$$\max_{\left\{k_{m,t}^{c}, n_{P,m,t}^{c,d}, n_{I,m,t}^{c,d}\right\}} F_{m,t}^{c}$$

subject to demand function (14). Real wages and the real rental rate of capital are then given by

$$r_{t}^{k^{c}} = mc_{t}z_{t}^{c}\gamma^{c} \left(k_{m,t}^{c}\right)^{\gamma^{c}-1} \left(\left(n_{P,m,t}^{c,d}\right)^{\alpha^{c}} \left(n_{I,m,t}^{c,d}\right)^{1-\alpha^{c}}\right)^{1-\gamma^{c}},$$
$$w_{P,t}^{c} = mc_{t}z_{t}^{c} \left(1-\gamma^{c}\right) \alpha^{c} \left(k_{m,t}^{c}\right)^{\gamma^{c}} \left(\left(n_{P,m,t}^{c,d}\right)^{\alpha^{c}} \left(n_{I,m,t}^{c,d}\right)^{1-\alpha^{c}}\right)^{(-\gamma^{c})} \left(\frac{n_{P,m,t}^{c,d}}{n_{I,m,t}^{c,d}}\right)^{\alpha^{c}-1},$$

and

$$w_{I,t}^{c} = mc_{t}z_{t}^{c}\left(1 - \gamma^{c}\right)\left(1 - \alpha^{c}\right)\left(k_{m,t}^{c}\right)^{\gamma^{c}} \left(\left(n_{P,m,t}^{c,d}\right)^{\alpha^{c}}\left(n_{I,m,t}^{c,d}\right)^{1 - \alpha^{c}}\right)^{(-\gamma^{c})} \left(\frac{n_{P,m,t}^{c,d}}{n_{I,m,t}^{c,d}}\right)^{\alpha^{c}},$$

where mc_t is the firm's real marginal cost. From the optimality conditions, all firms m face the same prices of factors, and since they have access to the same technology, marginal cost is equal across all firms at every period t.

Prices The Lagrangian for wages optimization problem takes the following form:

$$L = E_{t} \sum_{s=0}^{\infty} \left(\beta_{P} \xi^{p^{c}}\right)^{s} \epsilon_{t+s}^{b} \frac{\lambda_{P,t+s}^{c}}{\lambda_{P,t}^{c}} \frac{P_{t}^{c}}{P_{t+s}^{c}} \left\{ \begin{bmatrix} \tilde{P}_{t}^{c} \left(\frac{\tilde{P}_{t}^{c}}{P_{t+s}^{c}}\right)^{-\theta^{c}} y_{t+s}^{c} - r_{t+s}^{k^{c}} k_{m,t+s}^{c} \\ - w_{P,t+s}^{c} n_{P,m,t+s}^{c,d} - w_{I,t+s}^{c} n_{I,m,t+s}^{c,d} \end{bmatrix} + MC_{m,t+s} \times \\ \begin{bmatrix} z_{t+s}^{c} \left(k_{m,t+s}^{c}\right)^{\gamma^{c}} \left(\left(n_{P,m,t+s}^{c,d}\right)^{\alpha} \left(n_{I,m,t+s}^{c,d}\right)^{1-\alpha}\right)^{1-\gamma^{c}} \\ - \left(\frac{\tilde{P}_{t}^{c}}{P_{t+s}^{c}}\right)^{-\theta^{c}} y_{t+s}^{c} \end{bmatrix} \right\}.$$

To maximize the expected present value of their real dividends, the producers of intermediate goods in the consumption sector must meet the following first-order necessary condition with respect to \tilde{P}_t^c :

$$E_{t} \sum_{s=0}^{\infty} \left(\beta_{P} \xi^{p^{c}}\right)^{s} \epsilon_{t+s}^{b} \frac{\lambda_{P,t+s}^{c}}{\lambda_{P,t}^{c}} \left(\frac{\tilde{P}_{t}^{c}}{P_{t}^{c}}\right)^{-\theta^{c}} \prod_{k=1}^{s} \left(\frac{\left(\pi^{c}\right)^{\iota^{p^{c}}} \left(\pi_{t+k-1}^{c}\right)^{1-\iota^{p^{c}}}}{\pi_{t+k}^{c}}\right)^{-\theta^{c}} y_{t+s}^{c} \\ \left[\frac{\theta^{c}-1}{\theta^{c}} \left(\frac{\tilde{P}_{t}^{c}}{P_{t}^{c}}\right) \prod_{k=1}^{s} \left(\frac{\left(\pi^{c}\right)^{\iota^{p^{c}}} \left(\pi_{t+k-1}^{c}\right)^{1-\iota^{p^{c}}}}{\pi_{t+k}^{c}}\right) - mc_{t+s}\right] = 0,$$

with $\beta_P^s \epsilon_{t+s}^b \frac{\lambda_{P,t+s}^c}{\lambda_{P,t}^c} \frac{P_t^c}{P_{t+s}^c} MC_{t+s}^j$ being the Lagrange multiplier on demand function (14), and MC_{t+s}^j is the firm's nominal marginal cost. Since firms are assumed to act in the best interest of their owners (i.e. the lenders), the Lagrange multiplier is the marginal rate of substitution for consumption goods over time (i.e. Equation (47)). According to this expression, optimizing firms set nominal prices so that average future expected marginal revenues equate average future expected marginal costs. The expression above does not have a direct recursive formulation, making the computation difficult. However, writing the price-setting equation in recursive form eases this process. To do so, we need to define intermediate variables $x_t^{c,1}$ and $x_t^{c,2}$:

$$\begin{split} x_{t}^{1} &= \epsilon_{t}^{b} \frac{\theta^{c} - 1}{\theta^{c}} \tilde{p}_{t}^{c^{1-\theta^{c}}} y_{t}^{c} + \beta_{P} \xi^{p^{c}} E_{t} \left[\frac{\lambda_{P,t+1}^{c}}{\lambda_{P,t}^{c}} \left(\frac{\left(\pi^{c}\right)^{\iota^{p^{c}}} \left(\pi_{t}^{c}\right)^{1-\iota^{p^{c}}}}{\pi_{t+1}^{c}} \right)^{1-\theta^{c}} \left(\frac{\tilde{p}_{t}^{c}}{\tilde{p}_{t+1}^{c}} \right)^{1-\theta^{c}} x_{t+1}^{1} \right], \\ x_{t}^{2} &= \epsilon_{t}^{b} \tilde{p}_{t}^{c^{-\theta^{c}}} y_{t}^{c} m c_{m,t} + \beta_{P} \xi^{p^{c}} E_{t} \left[\frac{\lambda_{P,t+1}^{c}}{\lambda_{P,t}^{c}} \left(\frac{\left(\pi^{c}\right)^{\iota^{p^{c}}} \left(\pi_{t}^{c}\right)^{1-\iota^{p^{c}}}}{\pi_{t+1}^{c}} \right)^{-\theta^{c}} \left(\frac{\tilde{p}_{t}^{c}}{\tilde{p}_{t+1}^{c}} \right)^{-\theta^{c}} x_{t+1}^{2} \right], \\ x_{t}^{1} &= x_{t}^{2}. \end{split}$$

A.2.2 Housing Sector

The nominal profits (i.e. dividends) of the firm are denoted by

$$F_t^h = Q_t^h y_t^h - R_t^{k^h} u_t^{k^h} k_{t-1}^h - R_t^l l_{t-1} - W_{P,t}^h n_{P,t}^{h,d} - W_{I,t}^h n_{I,t}^{h,d}.$$

The firm's objective is a static problem of profit maximization

$$\max_{\left\{u_t^{k^h}k_{t-1}^h, n_{P,t}^{h,d}, n_{I,t}^{h,d}\right\}}F_t^h$$

subject to (18). Real wages and real rental rates of capital and land are then given by

$$\begin{split} r_{t}^{k^{h}} &= q_{t}^{h} z_{t}^{h} \gamma^{h} \left(u_{t}^{k^{h}} k_{t-1}^{h} \right)^{\gamma^{h}-1} l_{t-1}^{\gamma^{l}} \left(\left(n_{P,t}^{h,d} \right)^{\alpha^{h}} \left(n_{I,t}^{h,d} \right)^{1-\alpha^{h}} \right)^{1-\gamma^{h}-\gamma^{l}}, \\ r_{t}^{l} &= q_{t}^{h} z_{t}^{h} \gamma^{l} \left(u_{t}^{k^{h}} k_{t-1}^{h} \right)^{\gamma^{h}} l_{t-1}^{\gamma^{l}-1} \left(\left(n_{P,t}^{h,d} \right)^{\alpha^{h}} \left(n_{I,t}^{h,d} \right)^{1-\alpha^{h}} \right)^{1-\gamma^{h}-\gamma^{l}}, \\ w_{P,t}^{h} &= q_{t}^{h} z_{t}^{h} \left(1-\gamma^{h}-\gamma^{l} \right) \alpha^{h} \left(u_{t}^{k^{h}} k_{t-1}^{h} \right)^{\gamma^{h}} l_{t-1}^{\gamma^{l}} \times \\ \left(\left(n_{P,t}^{h,d} \right)^{\alpha^{h}} \left(n_{I,t}^{h,d} \right)^{1-\alpha^{h}} \right)^{-\gamma^{h}-\gamma^{l}} \left(\frac{n_{P,t}^{h,d}}{n_{I,t}^{h,d}} \right)^{\alpha^{h}-1}, \end{split}$$

$$w_{I,t}^{h} = q_{t}^{h} z_{t}^{h} \left(1 - \gamma^{h} - \gamma^{l}\right) \left(1 - \alpha^{h}\right) \left(u_{t}^{k^{h}} k_{t-1}^{h}\right)^{\gamma^{h}} l_{t-1}^{\gamma^{l}} \times \left(\left(n_{P,t}^{h,d}\right)^{\alpha^{h}} \left(n_{I,t}^{h,d}\right)^{1-\alpha^{h}}\right)^{-\gamma^{h} - \gamma^{l}} \left(\frac{n_{P,t}^{h,d}}{n_{I,t}^{h,d}}\right)^{\alpha^{h}}.$$

A.2.3 Labour Input

The firm's objective is a static problem of cost minimization

$$\min_{\left\{n_{P,e,t}^{j,d}, n_{I,e,t}^{j,d}\right\}} \int_{0}^{1} W_{P,e,t}^{j} n_{P,e,t}^{j,d} de + \int_{0}^{1} W_{I,e,t}^{j} n_{I,e,t}^{j,d} de.$$

A.3 Financial Intermediaries

To maximize the expected present value of their real dividends, financial intermediaries must solve

$$\max_{\{d_t,m_t\}} E_t \sum_{s=0}^{\infty} \left(\beta_P\right)^s \epsilon_{t+s}^b \left[\frac{\lambda_{P,t+s}^c}{\lambda_{P,t}^c} \frac{P_t^c}{P_{t+s}^c}\right] F_{t+s}^{fi}$$

subject to their balance sheet.

A.4 Competitive Equilibrium

An (imperfectly) competitive equilibrium is an allocation for:

- the lenders: $C_P = \left\{ c_{P,t}, h_{P,t}, n_{P,t}^j, b_{P,t}, i_t^{k^j}, k_t^j, u_t^{k^j}, d_t \right\}_{t=0, j \in \{c,h\}}^{\infty},$
- the borrowers: $C_I = \{c_{I,t}, h_{I,t}, n_{I,t}^j, b_{I,t}, m_t\}_{t=0, j \in \{c,h\}}^{\infty}$
- the firms in consumption sector: $\mathcal{F}^c = \left\{ y_{m,t}^c, K_{m,t}^c, n_{i,m,t}^{c,d}, F_{m,t}^c \right\}_{t=0,m\in[0,1],i\in\{P,I\}}^{\infty}$
- the firms in housing sector: $\mathcal{F}^h = \left\{ y_t^h, k_t^h, n_{i,t}^{h,d}, F_t^h, l_t \right\}_{t=0, i \in \{P,I\}}^{\infty}$, and
- prices system: $\mathcal{P} = \left\{ R_t, R_t^m, R_t^d, \pi_t^c, \tilde{p}_t^c, q_t^h, q_t^l, w_{i,t}^j, \tilde{w}_{i,t}^j, q_t^{k^j} \right\}_{t=0, i \in \{P,I\}, j \in \{c,h\}}^{\infty}$

and

such that, given initial conditions on predetermined variables, the exogenous processes, and the prices system, the allocations C_P , C_I , \mathcal{F}^c and \mathcal{F}^h solve the households' and firms' problems, and all market clearing conditions in Section 2.6 are satisfied.

B Definitions and Data Sources

Consumption (c_t)

$1983Q3\ to\ 2014Q4\ data$

Real (chained 2007 dollars) household consumption expenditure on non-durable goods, semi-durable goods, durable goods, and services per capita (number of persons of working age, 15 years and over), seasonally adjusted at annual rates. We remove the trend and isolate the cycle by applying a one-sided HP filter.

Source: Statistics Canada (Cansim Tables 282-0001 and 380-0064), Internal Calculations

Core CPI inflation rate (π_t^c)

1983Q3 to 2014Q4 data

All-items CPI excluding eight of the most volatile components and the core CPI. We splice both series, compute the annualized quarterly growth rate and remove the Bank of Canada's inflation target of 2 percent.

Source: Statistics Canada (Table 326-0020), Internal Calculations

Residential investment (y_t^h)

1983Q3 to 2014Q4 data

Real (chained 2007 dollars) business gross fixed capital formation in residential structures per capita (number of persons of working age, 15 years and over), seasonally adjusted at annual rates. We remove the trend and isolate the cycle by applying a one-sided HP filter.

Source: Statistics Canada (Cansim Tables 282-0001 and 380-0064), Internal Calculations

House price (π_t^h)

1983Q3 to 2014Q4 data

Real (core CPI) house price. We remove the trend and isolate the cycle by applying a one-sided HP filter. Source: Multiple Listing Service (MLS)

Non-residential investment (i_t^k)

1983Q3 to 2014Q4 data

Real (chained 2007 dollars) business gross fixed capital formation in non-residential structures, machinery and equipment per capita (number of persons of working age, 15 years and over), seasonally adjusted at annual rates. We remove the trend and isolate the cycle by applying a one-sided HP filter.

Source: Statistics Canada (Cansim Tables 282-0001 and 380-0064), Internal Calculations

Capital price (π_t^k)

1983Q3 to 2014Q4 data

Real (core CPI) implicit price index of business gross fixed capital formation in nonresidential structures, machinery and equipment. We remove the trend and isolate the cycle by applying a one-sided HP filter.

Source: Statistics Canada (Cansim Table 380-0066), Internal Calculations

Mortgage debt $(b_{i,t})$

1983Q3 to 2014Q4 data

Real (core CPI) residential mortgage credit per capita (number of persons of working age, 15 years and over), seasonally adjusted. We remove the trend and isolate the cycle by applying a one-sided HP filter.

Source: Statistics Canada (Cansim Tables 282-0001 and 176-0069), Internal Calculations

Nominal short-term interest rate (R_t)

1983Q3 to 2014Q4 data

Treasury bills rate, 3-months. We remove a linear time trend. Source: Statistics Canada (Cansim Table 176-0043), Internal Calculations

Nominal long-term interest rate (R_t^m)

1983Q3 to 2014Q4 data

Average residential mortgage lending rate, 5 years. We remove a linear time trend. Source: Statistics Canada (Cansim Table 176-0043), Internal Calculations

Hours worked in consumption sector (n_t^c)

1983Q3 to 2014Q4 data

Hours worked in the consumption sector per capita (number of persons of working age, 15 years and over). The full computation methodology for this series is available upon request. We remove the trend and isolate the cycle by applying a one-sided HP filter.

Source: Statistics Canada (Cansim Tables 281-0001, 281-0002, 281-0023, 281-0026 and 282-0001), Internal Calculations

Wage in consumption sector $(\pi_t^{w^c})$

1983Q3 to 2014Q4 data

Real (core CPI) wages in the consumption sector. The full computation methodology for this series is available upon request. We remove the trend and isolate the cycle by applying a one-sided HP filter.

Source: Statistics Canada (Cansim Tables 281-0004, 281-0029 and 281-0031), Internal Calculations

Capacity utilization rate in consumption sector $(u_t^{k^c})$

1983Q3 to 2014Q4 data

Capacity utilization rate in the consumption sector. The full computation methodology for this series is available upon request. We remove the mean.

Source: Statistics Canada (Cansim Tables 028-0001, 028-0002, 031-0005 and 031-0006), Internal Calculations

Hours worked in housing sector (n_t^h)

1983Q3 to 2014Q4 data

Hours worked in the housing sector per capita (number of persons of working age, 15

years and over). The full computation methodology for this series is available upon request. We remove the trend and isolate the cycle by applying a one-sided HP filter. Source: Statistics Canada (Cansim Table 281-0001, 281-0002, 281-0023, 281-0026 and 282-0001), Internal Calculations

Wage in housing sector $(\pi_t^{w^h})$

1983Q3 to 2014Q4 data

Real (core CPI) wages in the housing sector. The full computation methodology for this series is available upon request. We remove the trend and isolate the cycle by applying a one-sided HP filter.

Source: Statistics Canada (Cansim Tables 281-0004, 281-0029 and 281-0031), Internal Calculations

Capacity utilization rate in housing sector $(u_t^{k^h})$

1983Q3 to 2014Q4 data

Capacity utilization rate in the housing sector. The full computation methodology for this series is available upon request. We remove the mean.

Source: Statistics Canada (Cansim Tables 028-0001, 028-0002, 031-0005 and 031-0006), Internal Calculations